



This is an electronic reprint of the original article. This reprint may differ from the original in pagination and typographic detail.

Lönnroth, Tea; Krigsholm, Pauliina; Falkenbach, Heidi; Oikarinen, Elias

# Advancing understanding of the linkages between local land policy interventions and the responsiveness of housing supply : Intervention mechanisms in the Finnish context

Published in: Land Use Policy

DOI: 10.1016/j.landusepol.2024.107157

Published: 01/06/2024

*Document Version* Publisher's PDF, also known as Version of record

Published under the following license: CC BY

Please cite the original version:

Lönnroth, T., Krigsholm, P., Falkenbach, H., & Oikarinen, E. (2024). Advancing understanding of the linkages between local land policy interventions and the responsiveness of housing supply : Intervention mechanisms in the Finnish context. *Land Use Policy*, *141*, Article 107157. https://doi.org/10.1016/j.landusepol.2024.107157

This material is protected by copyright and other intellectual property rights, and duplication or sale of all or part of any of the repository collections is not permitted, except that material may be duplicated by you for your research use or educational purposes in electronic or print form. You must obtain permission for any other use. Electronic or print copies may not be offered, whether for sale or otherwise to anyone who is not an authorised user.

ELSEVIER



### Land Use Policy



journal homepage: www.elsevier.com/locate/landusepol

### Advancing understanding of the linkages between local land policy interventions and the responsiveness of housing supply: Intervention mechanisms in the Finnish context

Tea Lönnroth<sup>a,\*</sup>, Pauliina Krigsholm<sup>a,b</sup>, Heidi Falkenbach<sup>a</sup>, Elias Oikarinen<sup>a,c,d</sup>

<sup>a</sup> Aalto University, Department of Built Environment, Real Estate Research Group, P.O. Box 14100, 00076 Aalto, Finland

<sup>b</sup> Finnish Geospatial Research Institute, National Land Survey of Finland, Vuorimiehentie 5, Espoo 02150, Finland

<sup>c</sup> University of Oulu, Department of Economics, Accounting and Finance, University of Oulu, P.O. Box 8000, FI-90014, Finland

<sup>d</sup> University of Turku, Department of Economics, FI-20014 Turun yliopisto, Finland

#### ARTICLE INFO

Keywords: Land policy interventions Land use regulation Intervention mechanisms Housing supply

#### ABSTRACT

Land policy interventions are important ways for municipalities to implement their spatial development objectives. Such interventions may, however, generate both intended and unintended effects to housing supply and its responsiveness. It is therefore essential to better understand how and through what kind of mechanisms local land policy interventions can influence the quantity of housing supply. This paper addresses these questions. First, we develop a conceptual classification of so-called intervention mechanisms that describes the linkages between local land policy interventions and the quantity and responsiveness of housing supply. The intervention mechanisms build on a review of the urban economics literature and are further developed empirically using extensive expert interview data on local land policy interventions and practices in 30 Finnish municipalities. Based on the understanding that such intervention primarily influences housing supply hrough either direct or indirect costs on development, we conceptualize seven intervention mechanisms in the Finnish land policy context. Second, we demonstrate the variation in the application of land policy interventions that contribute to the (in)responsiveness of housing supply at the municipal level. Our findings highlight significant variation in the application of such interventions, potentially leading to different outcomes in housing supply restrictiveness. The findings contribute to a deeper understanding of the mechanisms by which land policy interventions operate to affect the (in)responsiveness of housing supply.

#### 1. Introduction

Land policy interventions, the deliberate actions and measures initiated by local governments (often municipalities) to shape, influence, or direct land use and housing development, are important ways for achieving various environmental, social, and economic goals (see e. g., Debrunner and Hartmann, 2020; Gerber et al., 2018b; Hartmann and Spit, 2015; Shahab et al., 2021; Vejchodská et al., 2022). These interventions encompass – besides classic planning measures – all different types of regulatory measures and incentives related particularly to the implementation process of spatial development objectives (Debrunner and Hartmann, 2020; Gerber et al., 2018b; Vejchodská et al., 2022).<sup>1</sup>

Land policy interventions serve an important public purpose by

reducing negative externalities (e.g., congestion, noise, and environmental impacts) that would occur in an unregulated market (Hilber and Vermeulen, 2016; Schill, 2005). They can also be used to advance different positive externalities, such as the provision of open space and building safety (e.g., May, 2003; Wang, 2014; McFarlane et al., 2021), as well as to facilitate plan implementation (e.g., Turnbull, 1988; McFarlane, 1999; Ihlanfeldt, 2004; Lyytikäinen, 2009). However, these interventions may also create undesirable subsidiary effects that are unanticipated by policymakers (Brueckner, 2009; Shahab et al., 2019). Sometimes these subsidiary effects may even generate costs that offset the benefits originally intended (Turner et al., 2014). For instance, interventions aimed to preserve open space have been found to contribute to a more sprawled development pattern (Irwin and Bockstael, 2004),

https://doi.org/10.1016/j.landusepol.2024.107157

Received 14 September 2022; Received in revised form 8 March 2024; Accepted 22 March 2024 Available online 6 April 2024

0264-8377/© 2024 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

<sup>\*</sup> Correspondence to: Aalto University, Department of Built Environment, P.O. Box 14100, 00076 Aalto, Finland.

E-mail address: tea.lonnroth@aalto.fi (T. Lönnroth).

<sup>&</sup>lt;sup>1</sup> Following Alexander (2014), land policy interventions exclude thus, for example, fiscal policies and banking regulation that are not directly involved with and affected by planning and plan implementation.

whereas the imposition of inclusionary housing has been found to reduce housing supply and increase the price of market-rate housing (Powell and Stringham, 2004; Bento et al., 2009; Schuetz et al., 2011).

While policymakers should carefully consider the potential subsidiary effects of their interventions to minimize negative consequences, the inherent complexity and interconnectedness of different interventions and their outcomes make this a challenging task. This challenge seems particularly relevant when considering the effects of land policy interventions on housing supply and its responsiveness. The responsiveness of housing supply reflects how quickly and effectively the supply responds to demand, and it thereby plays a pivotal role in determining the resulting quantity of housing supply - a key focus of this paper. Delving into the linkages between land policy interventions and the supply-side of housing is important for various reasons. Foremost, the size of housing stock significantly influences housing prices and rents (Cheshire and Vermeulen, 2009; Hilber and Vermeulen, 2016; Albouy and Ehrlich, 2018; Anthony, 2006; 2017). Supply constraints on housing markets may cause a higher house price-income ratio, and thus contribute to housing unaffordability (Oikarinen et al., 2023). Moreover, the responsiveness of (quantity of) housing supply has been found to influence, for instance, regional growth dynamics, labour market flexibility, migration, and income levels (Glaeser et al., 2006; Saks, 2008; Zabel, 2012; Diamond, 2017; Ganong and Shoag, 2017; Hsieh and Moretti, 2019; Accetturo et al., 2021).

Despite the growing recognition of the profound effects of the supply-side factors in housing markets, there remains a lack of clarity on the linkages between land policy interventions and the quantity of housing supply at the local government level. Earlier studies have mainly focused on categorizing planning-related regulatory barriers and examining their influence on housing construction and prices in the US and UK contexts (e.g., Bramley and Watkins, 2014; Glaeser and Ward, 2009; Gyourko et al., 2008; Ihlanfeldt, 2007; Jackson, 2016; Anthony, 2006; 2017). Much less attention has been devoted to examining the linkages of other types of land policy interventions, including those related to plan implementation, i.e., the way land is distributed for future (residential) development and how this development process is enforced (see e.g., Krigsholm et al., 2022). Such interventions are particularly relevant in countries where both the spatial planning and the land development process are strongly controlled by local public authorities, as is the case in many European countries (Valtonen et al., 2017a).

The aim of this paper is to advance understanding of the interrelations between land policy interventions and the quantity of market-rate housing supply.<sup>2</sup> In this context, we refer to *land policy interventions* as all those actions and measures initiated by local governments to shape, influence, or direct land use and future housing development to advance the (politically defined) spatial development objectives. Specifically, such interventions encompass a variety of (regulatory) measures and incentives including, for instance, strategic land banking and land acquisition practices, local planning requirements, and conveyance stipulations of buildable plots.

The first objective is to develop a conceptual classification that outlines various mechanisms through which land policy interventions may influence the quantity of housing supply at the housing market level. To that end, we draw on the economic principles within the context of urban economics literature to demonstrate the potential linkages between such interventions and the size and growth of housing stock. According to this strand of the literature, local public authorities may essentially influence the quantity of housing supply either directly through land availability, restrictions on housing supply, and development costs, or indirectly through time-related indirect costs (e.g., Mayer and Somerville, 2000; Paciorek, 2013; Skidmore and Peddle, 1998). These two linkages enable us to identify various interventions and practices of local land policy that contribute to the (in)responsivenessand thereby to the quantity – of housing supply. Further, we utilize extensive interview data on land policy interventions and practices from the 30 most populated municipalities in Finland, a country with a tradition of strong interventions by local public authorities on the land market (see e.g., Krigsholm et al., 2022; Valtonen et al., 2017a). This empirical approach allows us to provide more detailed insights into these two linkages. Specifically, by identifying and grouping together those interventions and practices with an identified influence either directly or indirectly on the costs of development, we conceptualize a total of seven intervention mechanisms.

As the ways in which municipalities activate different land policy instruments vary greatly (Shahab et al., 2021; Krigsholm et al., 2022), also the prevalence and intensity of interventions influencing the quantity of housing supply is likely to be heterogenous at local government level. Hence, the second objective of the study is to identify and demonstrate the variations in the application of housing-supply relevant land policy interventions at the local government level. Furthermore, we qualitatively rank the alternative applications of these interventions from the vantage-point of market-rate housing supply. By doing so, this study does not attempt to evaluate or judge the public objectives of land policy interventions, nor does it aim to quantify the effect of land policy choices on the (in)responsiveness of housing supply. Instead, the focus is to explore how the quantity of housing supply is influenced by local land policy interventions. Ultimately, our methodology showcases a structured way to capture how the different choices of municipalities can pile up into different degrees of housing supply restrictiveness of local land policy.

The paper is structured as follows. Section 2 reviews the existing literature that links land policy interventions to the quantity of housing supply and its responsiveness to lay the basis for conceptualizing intervention mechanisms. Section 3 describes the research process, the institutional framework, and the methods of data collection and analysis. In Section 4 we present the intervention mechanisms and the variation in the application of those land policy interventions that contribute to the (in)responsiveness of housing supply in the Finnish context. Section 5 discusses the findings and Section 6 concludes.

#### 2. Linking land policy interventions and housing supply

In this section, we review the theoretical and empirical literature to develop an understanding of the linkages between land policy interventions and the quantity of housing supply. Specifically, our focus is on the urban economics literature that addresses land policy interventions influencing the size and growth of housing stock. Given that the extant empirical literature on land policy interventions has mainly concentrated on planning-related interventions, the review of the literature is strongly guided by this subset. While not exhaustive, the review establishes an understanding of the basic mechanisms through which land policy interventions influence the quantity of housing supply (henceforth referred to as 'housing supply' unless mentioned otherwise). We start by presenting theoretical considerations of housing supplyrelated effects from such interventions, after which we present empirical evidence on the topic.

#### 2.1. Theoretical considerations

Economic theory suggests that the quantity of housing supply and the price of housing are dependent on the interaction between supply and demand components of the market. In a competitive market, supply and demand intersect at an equilibrium. When either demand or supply schedule shifts, for example due to land policy interventions, the

<sup>&</sup>lt;sup>2</sup> We refer to market-rate housing as non-subsidized housing. While we acknowledge that market-rate housing will not necessarily address the housing needs of all households, it is most likely to also benefit lower-income households (see e.g., Bratu et al., 2023; Mast, 2021; Been et al., 2019), making it an important factor to consider.

equilibrium will likewise change. On the demand side, these interventions can provide amenities that lead to increased local housing demand and, consequently, higher house prices (Anthony, 2017; Quigley and Rosenthal, 2005). Conversely, they can also create disamenities, resulting in a decrease in housing demand and lower prices.

On the supply-side, land policy interventions can influence, for instance, the permitted development intensity, the cost of development, or the amount of land available for housing development. Whether these interventions are regarded as planning-related or fall into other types of land policy interventions can depend on the institutional context. Hence, the classification of these interventions may vary based on the prevailing institutional framework, making the distinction sensitive to the specific context in which they are applied. Regardless of whether an intervention falls under planning or under other land policy intervention, the basic mechanisms remain the same. Ultimately such interventions impact the housing supply-schedule and thus the equilibrium outcome (Gyourko et al., 2008; Mayo and Sheppard, 1996).

The supply-side effect can be modelled in two different ways. The first way is to consider an underlying supply curve, and model land policy interventions as items that shift the supply curve up (down) through increased (decreased) development costs. When these interventions increase the developers (marginal) costs for housing development, the effect on new housing supply follows the basic idea of economic theory: based on the assumption of a competitive market, a cost imposed by the public authorities will shift up the supply curve of housing (from the Supply<sub>0</sub> curve to the dashed Supply<sub>1</sub> curve presented in Fig. 1), resulting in a lower quantity of new housing built and hence a lower housing supply and a higher price level (point B vs. point A) (Evans, 2004; Huffman et al., 1988; Been, 2005).

The ultimate effect of land policy interventions on housing supply will, however, depend not only on the actual cost imposed by such interventions but also on the likelihood that some of the cost will be capitalized into house prices or the price of residential land. If the intervention simultaneously creates a benefit for the homebuyer, we would expect it to be reflected in a higher house price (see e.g., Anthony, 2017). Consequently, if the cost imposed by land policy interventions is fully reflected in the house price, developers would not experience any burden from the increased costs, and there would be no decrease in the number of homes.<sup>3</sup>

However, if homebuyers value the benefits of land policy interventions less than the cost imposed, the increase in the house price would be insufficient to cover the cost. In such cases, the higher marginal cost of development would reduce the price developers are willing to pay for the land (see e.g., Brueckner, 2009). Also, the uncertainties faced by developers regarding the capitalization of costs into the house price could similarly reduce the price of land. By doing so, land policy interventions lower the opportunity cost of using the land for alternative purposes. This "reservation price" problem (see, for example, Nelson, 1994; Huffman et al., 1988) may result in land being withheld from residential use for more extended periods. After all, land has a variety of uses beyond housing, including not only agricultural purposes but also other urban purposes that already generate income or benefit to its owner (Metcalf, 2018).

Consequently, when landowners reduce the rate at which they supply residential land, the housing development process is slowed down, leading to a reduced number of homes (Powell and Stringham, 2004; Brueckner, 2000). The capitalization of the costs and the subsequent impact of interventions on housing supply vary not only depending on the type of intervention but also on the specific context, such as the institutional context that affects the structure of the land market (Needham, 2000). Addressing this, though valuable, is an empirical question which falls outside the scope of this paper.

While some land policy interventions might decrease housing supply, they can naturally also work in the opposite direction by supporting or even enhancing it. For example, public authorities can employ these interventions to plan and supply more land for residential development than what an unregulated market would provide (Ihlanfeldt, 2004). This would reduce the marginal cost of development, shift the supply curve downwards, and thereby increase the overall supply of housing at any given price. Public authorities can also increase housing supply by introducing various supply subsidies, such as economic incentives or allowances, to impact the timing decisions of developers and thus encourage them to provide housing (McFarlane, 1999; Turnbull, 1988).

The second way to model the influence of land policy interventions on the supply side is through their influence on the (in)responsiveness of housing supply to price changes, i.e., on the price elasticity of housing supply, which defines the slope of the supply curve (see Fig. 1). By influencing the marginal cost of development, land policy interventions can thereby affect the elasticity of housing supply. This applies, for example, to interventions that influence the costs of building materials, impose height restrictions, or affect the time required to obtain a building permit (Henderson, 2009). Assuming all other things equal, we would expect a municipality with relatively low levels of housing supply restrictions through land policy interventions to have a relatively elastic housing supply. A positive shift in demand would therefore result in a notable increase in the housing stock and relatively modest increase in housing costs (point A in Fig. 1). In contrast, if a municipality has higher levels of housing supply restrictions, the housing supply can increase less when faced with a positive demand shift. Instead, the positive shift in demand would lead to a greater increase in housing prices (point B in Fig. 1).

Regardless of whether we approach the influence of land policy interventions through changes in development costs or price elasticity of housing supply, we would often expect an increase in the marginal cost of housing development to result in a reduction in the quantity of housing supply.<sup>4</sup> Henceforth, we refer to this effect resulting from changes in the (marginal) cost of development as *(in)responsiveness of housing supply*.

There are various forms of land policy interventions that influence housing development and the use of land, ranging from planning and development restrictions to different interventions related to the implementation process of plans, including those interventions targeting the strategic acquisition and public management of land. From the perspective of marginal costs, they can ultimately be collapsed into two main categories (Mayer and Somerville, 2000; Henderson, 2009). The first includes those interventions that directly affect development costs, such as interventions that require more expensive building components, impose exactions, or restrict buildable land supplies. The second category involves various time-related interventions in the decision-making and development process. These interventions implicitly influence development costs, for example, by introducing multiple reviews in obtaining a building permit or prolonging the conveyance processes of publicly owned land.

#### 2.2. Direct effects of land policy interventions

One way for local governments to influence the (in)responsiveness of housing supply is by affecting the amount of land available for housing

<sup>&</sup>lt;sup>3</sup> It is relevant to note that even in these cases, if developers are required to pay the cost upfront, it could burden some smaller developers. This could potentially lead to those developers to exit the market, thereby making it less competitive (Anthony, 2003).

<sup>&</sup>lt;sup>4</sup> Although the cost can in some cases affect land prices, it will in many cases ultimately exert downward pressure on the quantity of (new) housing (see, for example, Brueckner, 2009). Moreover, in some cases, these interventions may also affect the timing decisions of development, which may lead to a (temporary) increase in the responsiveness of housing supply (see e.g., Titman, 1985). Such effects are addressed in the following subsections.



**Fig. 1.** The effect of land policy interventions on housing supply and prices. *Note*: The figure illustrates the effect of land policy interventions that increase the (marginal) cost of development on housing supply and prices (adapted from Henderson, 2009). It is important to note that the magnitude and effect of the cost imposed by land policy interventions on the quantity of housing supply may vary depending on the specific intervention type. In some cases, the cost may be capitalized into land prices, thus lowering the price developers are willing to pay for residential land. The cost would therefore affect the supply of residential land, and through land supply ultimately the number of homes.

construction through greenbelts, urban growth boundaries or other land use enactments that hinder residential land supply (e.g., Gerber et al., 2018a).<sup>5</sup> Numerous studies have shown that interventions to restrict land supply from development lead to higher land costs and thus impact the responsiveness of housing supply (Barlow, 1993; Caesar, 2016; Hannah et al., 1993; Levine, 1999; Monk and Whitehead, 1996; Pollakowski and Wachter, 1990; Kok et al., 2014). In the short run such interventions can, however, increase housing supply by reducing the uncertainty related to development (see e.g., Cunningham, 2007; Titman, 1985). It is relevant to note that restricted land supply may also diminish opportunities for spatial competition between developers and thus lead to monopoly power (Ball, 2003). Non-competitive market conditions allow developers greater scope to influence both housing output levels and land prices.

In contrast to land supply constraints that effectively increase the price of land, a variety of land policy interventions can also influence the (in)responsiveness of housing supply directly through impacts on the (marginal) cost of development. Such interventions are expected to reduce the quantity of housing supply unless the increased costs can be capitalized into decreased values of land (Kok et al., 2014; Schuetz et al., 2011). However, when land values decrease, it alters the opportunity cost of keeping the land for an alternative use (e.g., Brueckner, 2009; Metcalf, 2018; Ihlanfeldt and Shaughnessy, 2004). At lower prices, fewer landowners are thus willing to sell their land which imply lower levels of residential land supply and, consequently, reduced housing production (Powell and Stringham, 2004).

The most direct way of imposing costs on housing development is presumably through the imposition of different fees or exactions. Examining the effects of impact fees, Skidmore and Peddle (1998) found that housing development is significantly reduced in fee areas compared to similar municipalities with no such fees. On the other hand, while the imposition of impact fees increases development costs, it may at the same time (indirectly) reduce other costs related to project approval as well as reduce uncertainty through increased rates of project approval (Burge and Ihlanfeldt, 2006; Mayer and Somerville, 2000). Hence, the net effect of impact fees on costs can be negative, thus increasing housing supply. Moreover, impact fees may enable development that would not occur in an unregulated market. This might happen in situations where impact fees levied on developers enable the provision of infrastructure that would otherwise be stalled by the public authorities (Been, 2005).

Also, public interventions that affect the costs of development could influence the option value of waiting and thus incentivize building activity (Murray, 2022; Capozza and Li, 1994). For example, the Finnish three-rate property tax system allows, and in some areas even necessitates, undeveloped residential land to be taxed at a higher rate than developed land. The introduction of the tax was found to increase initial single-family housing starts by an average of 12% (Lyytikäinen, 2009). Similarly, Murray (2018) found that the announcement of changes in development costs affected the timing decisions of development, where higher costs encourage faster development. These findings signify developers' increased willingness to exercise their development options, although it should be noted that the long-run effects are likely smaller than the initial ones (e.g., Lyytikäinen, 2009).

Other types of land policy interventions that increase development costs, such as building codes that require the use of more expensive building components, may also act as constraints on housing supply (e. g., Listokin and Hattis, 2005; McFarlane et al., 2021). Different forms of development requirements, such as inclusionary housing obligations that require developers to either build or cede land for affordable housing (Muñoz Gielen et al., 2017), have likewise been found to contribute to lower rates of housing development (e.g., Bento et al., 2009; Powell and Stringham, 2004; Schuetz et al., 2011). Although such obligations may be politically attractive ways of promoting housing affordability, (see, for example, Bento et al., 2009), these housing units are in general sold at a price below market price implying that developers will receive lower revenues on those units. In response, developers may reduce the price they are willing to pay for residential land and therefore also the quantity of housing to be supplied (Schuetz et al., 2011). Moreover, inclusionary housing obligations also represent a tax

<sup>&</sup>lt;sup>5</sup> This by no means implies that such interventions are outright harmful. For example, green spaces may contribute to positive externalities and thus have benefits that mitigate their costs (see, for example, Cheshire and Sheppard, 2002). However, in this paper our focus is on linkages between land policy interventions and quantity of housing supply, and so we do not discuss such benefits further.

on the construction of market-rate housing (Glaeser, 2019) and just as with any other taxes, the cost is borne by some combination of market-rate homebuyers, landowners, and developers (Powell and Stringham, 2004). Hence, ultimately, economic theory suggests that inclusionary housing obligations act to decrease the total supply of housing (Bento et al., 2009). Another form of development requirements that has become increasingly prominent during the past decades is environmental regulations (see e.g., Schill, 2005). Examining the imposition of critical habitat designations, Zabel and Paterson (2006) found that such regulations result in significant decreases in the supply of housing permits. Their findings suggest that the proposal of critical habitat designations acts as a signal that development will be more costly and thus adversely impacts building activity.

The above discussion has mainly focused on land policy interventions and costs impacting housing development. The most extreme way of affecting new housing supply, however, is through concrete constraints on building. These can prohibit building altogether or impose caps on the number of permits or building volumes. When such constraints are binding, they lower the responsiveness of housing supply by effectively making the marginal cost of housing development infinite (Gyourko and Molloy, 2015). It should be noted, however, that while constraining housing supply, such interventions can also reduce market uncertainty and, consequently, the option value of land. This reduction in uncertainty can make immediate development relatively more attractive, thereby increasing the optimal rate of supply in the short run (see e.g., Murray, 2022).

#### 2.3. Indirect effects of land policy interventions

Moving on to less tangible cost-influencing interventions, one empirically widely recognized determinant of housing supply is the implicit costs associated with delays in the planning and development process, which may be even more harmful to new housing supply than the direct cost-impacting regulations (Mayer and Somerville, 2000; Ball, 2011; Paciorek, 2013; Wong et al., 2011). Any interventions that extend the approval times for development projects will reduce the probability of development (Wrenn and Irwin, 2015). Delays in the land policy system may, consequently, lead to a mismatch between demand and new housing development and reduce the housing stock available in all future points in time (Skidmore and Peddle, 1998).

The findings on the detrimental impacts of drawn-out development processes are supported by Ball (2011), who found that the overall time to receive planning permission in the UK is substantial and may play a significant role in generating low housing supply responsiveness. Paciorek (2013) documented a strong negative relationship between the permitting lags and the responsiveness of housing supply in the US. Furthermore, additional costs associated with bureaucratic uncertainty in the approval process have been found to be substantial enough to significantly reduce or displace new construction (Staley, 2001).

Next to public authorities, private landowners and developers themselves may also induce delays in the development process. Private landowners, for example, may delay or block land assembly by resisting to sell their land, thereby creating a holdout problem (Grossman et al., 2019; Miceli and Sirmans, 2007). To address and mitigate such market failures, public authorities may have the option to employ various interventions, such as expropriation, to facilitate a more efficient development process (Miceli and Sirmans, 2007). Correspondingly developers may voluntarily delay housing production and withhold land from the market through land banking in anticipation of future gains (Murray, 2020), with adverse effects on the relationship between land supply and housing supply (Tse, 1998). In a public land development context (see e.g., Needham, 1997), where the public authorities are heavily involved in the land market and where the bulk of housing development occurs on publicly supplied land, such speculative behaviour is, however, less prominent. Conversely, the land banking behaviour by public authorities may enable a closer match-up between supply and demand changes, as the public authorities may assign and release land for development purposes and thus alleviate delays (Barlow, 1993).

In summary, the reviewed literature suggests that land policy interventions may influence the (in)responsiveness of housing supply, and thereby the size of housing stock, by affecting the marginal cost of development (see Fig. 2). However, there is no single defining form of such interventions, and their impacts may arise through several different channels depending on the institutional arrangements. The notion that government interventions may influence housing development by affecting the (marginal) cost of development either through direct costs or time-related indirect costs is supported by the extant theoretical and empirical literature. This suggests that while there may be different costs from, for example, creating and using institutions (Buitelaar, 2004), the basic avenues linking land policy interventions and housing supply are often the same. These notions provide a basis for formulating different empirically driven mechanisms through which land policy interventions may influence the (in)responsiveness of housing supply.

#### 3. Research process and data collection

## 3.1. Conceptualizing intervention mechanisms and exploring variation in land policy interventions

Our empirical analysis consisted of two main phases (see Fig. 3). The first phase focused on the formulation of intervention mechanisms and involved two steps of analysis. In the first step, we focused on mapping the Finnish land policy environment. To that end, we employed extensive data collected in interviews with municipal representatives responsible for local land policy in the 30 most populated municipalities in the country. This resulted in a general list of 43 land policy instruments and practices (see Appendix A).

The second step of Phase 1 was based on a combination of deductive and inductive reasoning. Prior theoretical knowledge was used to refine the general list to cover only those instruments and practices with relevance to the responsiveness of housing supply (henceforth called land policy indicators). As explained in Section 2, public interventions may either aid or constrain the responsiveness housing supply by influencing the marginal costs of development (e.g., Glaeser and Gyourko, 2003; Brueckner, 2009; Ihlanfeldt, 2004). These can essentially be grouped into two avenues of basic mechanisms: direct cost-impacting measures, such as restrictions, requirements or allowances, and time-related indirect cost impacts, which can be materialized through delays or uncertainties in the development process. For the Finnish case, we identify a total of 25 indicators with either indirect or direct effects on the (in)responsiveness of supply.<sup>6</sup> These land policy indicators were further grouped together based on their distinctive cost-influencing features to formulate seven intervention mechanisms (see Fig. 4).

In Phase 2 of the research process, the objective was to identify differences in the application of land policy interventions across municipalities and to assess the variation from the perspective of housing supply restrictiveness. This phase of the analysis focused on developing a scaling scheme that reflects the current spectrum of local land policy practices for each of the identified indicators. The empirical data was analysed one indicator at a time, on the basis of which we developed an empirically grounded scaling scheme that qualitatively ranked the alternatives from the vantage-point of housing supply. Each scaling scheme is based on ordinal variables (three-point Likert scale), where lower score connotes a stricter intervention, i.e., the most

<sup>&</sup>lt;sup>6</sup> Some land policy instruments and practices were merged under one land policy indicator to reflect the data and the variation across municipalities in the best way possible. Thus, a selected indicator is not necessarily equivalent to a single land policy instrument or practice.



**Fig. 2.** Overview of the main linkages between land policy interventions and housing supply responsiveness. *Note*: The figure summarizes the effects through which various land policy interventions, including different types of regulatory measures, incentives, process delays, and planning tools, can influence the (in)responsiveness of housing supply. Note that while the figure describes the linkages between cost-inducing interventions and housing supply, land policy interventions may also work in the other direction by reducing the (marginal) costs of development. Additionally, it should be noted that certain types of land policy interventions, like impact fees, may benefit homebuyers without necessarily impacting housing supply responsiveness. Such effects are associated with the demand-side of the framework, and not modelled here. For a discussion on demand-side effects, see e.g., Anthony (2006); (2017).

disadvantageous or least beneficial mode of intervention from the perspective of supply responsiveness, depending on the type of land policy indicator and its expected way of influence. At this stage of the process, indicators with no variation were excluded from further analysis. As a result, the total number of land policy indicators with either direct or indirect effects was refined to 23.

#### 3.2. Institutional framework and data collection

This study focuses on the Finnish land use planning and policy system. The Finnish planning system is a statutory framework defined by the Land Use and Building Act (1999), which sets out general regulations and objectives for land use planning and policy. Municipalities are charged with creating and updating plans that allow building development within their jurisdictions. While landowners have the discretion to initiate land use plans, municipalities ultimately have the power to independently decide when and where planning processes are commenced. All such legally binding detailed plans are also approved by the municipalities. The development of a property requires a building permit issued by the municipality, and in general, permits can only be granted if there exists a legally binding plan that allows for the proposed development. However, the municipality has the right to decide not only on the content of the legally binding detailed plans, but also on any deviations from those plans.<sup>7</sup>

In Finland, local public interventions through land policy are extensive. Since 2015, the Land Use and Building Act has included a definition of land policy, which it specifies as one of the duties of municipalities (Land Use and Building Act, 1999, amendment 204/2015). As is the case in countries such as Sweden, the Netherlands and China, Finnish municipalities often engage in public land development (Needham, 1997; Valtonen et al., 2018; van der Krabben and Jacobs, 2013). Hence, municipal authorities are heavily involved in the land market by acquiring land, controlling the land development process, and acting as a main supplier of buildable land next to their role as a planning regulator (Buitelaar, 2010; Krigsholm et al., 2022; Needham, 1997). Municipalities also have the statutory right to use compulsory instruments, such as expropriation and pre-emption rights, to steer and facilitate land use and development within their jurisdiction (Buitelaar, 2010; Hartmann and Spit, 2015). However, the Finnish legislative framework provides for flexibility in the use of the available land policy instruments, depending on the spatial development objectives of the municipality.

<sup>&</sup>lt;sup>7</sup> See, for example, Valtonen et al. (2017b) for more on the Finnish planning system.



Fig. 3. The different phases and steps of the research process.



Fig. 4. Overview of the conceptual classification of intervention mechanisms.

To examine the ways in which Finnish municipalities intervene in land markets and housing construction, data was collected mainly through extensive interviews. The interview questions were constructed and chosen with a view to gaining a comprehensive insight into land policy decision-making and current practices in Finnish municipalities. The semi-structured interviews (e.g., Robson, 2002) covered more general themes such as approaches to land policy as well as more detailed questions on different land policy instruments and their implementation. The sample includes 30 of the 31 most populated Finnish municipalities with populations ranging from 37,000 to 650,000 in 2020.<sup>8</sup> The sample accounts for 61% of the 2020 Finnish population and over 70% of the number of housing units constructed in the period 2014–2020, making it highly representative of the land policy interventions and practices in Finnish urban municipalities. The

municipalities represented in the sample also provide comprehensive geographic coverage and demographic heterogeneity.

The interviews were carried out from March 2020 to August 2020 via remote connection. They targeted key personnel responsible for land policy actions in each municipality, such as the head of land surveying and land use unit, land use experts, or similar. These interviewees were expected to possess the best knowledge on local land policies and practices in their municipality. For the most part interviews were conducted with one person from each municipality, but in eight cases two or three persons were interviewed to secure answers to all questions. All interviews were recorded with the permission of the interviewees. The interviews ranged in length from 60 to 210 minutes, and all interviews were transcribed and manually coded using an inductive coding technique (e.g., Miles and Huberman, 1994). In the analysis stage, we identified missing data related to specific questions from nine municipalities. To complement our interview data with this regard, we recontacted the interviewees via email or phone and asked for necessary additional information. The interview data was further complemented

 $<sup>^{\</sup>rm 8}$  In total 31 municipalities were contacted, of which one declined the interview request.

with available municipal documents, such as land policy agendas and land use agreements.

#### 4. Intervention mechanisms and related land policy indicators

Building on the two main avenues of (marginal) cost-impacting factors discussed in Section 2, this paper formulated intervention mechanisms to aid the conceptual understanding of how land policy interventions may influence the (in)responsiveness of housing supply.

#### Table 1

| Intervention mechanisms and related land policy indicators influencing the (in) |
|---|
| responsiveness of housing supply.   |

|   | Intervention<br>mechanisms         | Land policy<br>indicators  | Expected way of<br>influence on<br>responsiveness and<br>quantity of housing<br>supply (+/-) |
|---|------------------------------------|--|--|
| Time-<br>related<br>indirect<br>effects | Municipal land<br>reserve building | Land acquisition<br>activity and land<br>holdings<br>Pre-emption of land | + +  |
|   | Delay in planning                  | Expropriation<br>Stakeholder   | +  |
|   |                                    | involvement<br>Stakeholder   | +/-  |
|   |                                    | pressure<br>Planning delay   | -  |
|   |                                    | (incl. new plans<br>and alterations)                                     |  |
|   |                                    | Plan appeals   | -  |
|   |                                    | (Weak) State of  | -  |
|   |                                    | municipal finance  |  |
|   | Delay in                           | Conveyance   | -  |
|   | availability of<br>buildable land  | process delay  |  |
|   |                                    | Planning of<br>privately owned   | +/-  |
|   |                                    | land for new   |  |
|   |                                    | residential areas  |  |
|   | Delay in plan                      | Post-planning  | -  |
|   | implementation                     | process<br>Conveyance  |  |
|   |                                    | stipulation of   |  |
|   |                                    | building obligation<br>(extensions)                                      |  |
|   |                                    | Reminder to build  | +  |
| Direct<br>effects                       | Production costs<br>and impact on  | Land value capture<br>practices  | +/-  |
|   | profit margin                      |  |  |
|   |                                    | Incentives in land<br>use agreements                                     | +  |
|   |                                    | Inclusionary   | -  |
|   |                                    | housing  |  |
|   |                                    | requirements   |  |
|   |                                    | Residential<br>structure   | -  |
|   |                                    | requirements<br>Requirements of  | _  |
|   |                                    | local detailed plans   |  |
|   |                                    | Objects of   | -  |
|   |                                    | protection   |  |
|   | Market<br>competition              | Developer steering   | +/-  |
|   |                                    | Land market<br>constraints   | -  |
|   | Builders' cost of                  | Applicability of   | +  |
|   | option to delay                    | contractual  |  |
|   |                                    | penalties<br>Vacant urban land<br>tax                                    | +  |
|   |                                    | un la                                |  |

Land Use Policy 141 (2024) 107157

As outlined in Table 1, we suggest that the interventions influencing the supply (in)responsiveness can be categorized into seven intervention mechanisms: 1) Municipal land reserve building, 2) Delay in planning, 3) Delay in availability of buildable land, 4) Delay in plan implementation, 5) Production costs and impact on profit margin, 6) Market competition and 7) Builders' cost of option to delay. Based on the theoretical and empirical linkages between land policy interventions and housing supply outlined in Section 2, Table 1 also reports whether the land policy intervention or practice is expected to have an increasing or decreasing effect on the responsiveness of housing supply.

The formulation of intervention mechanisms emanates from 23 land policy indicators in the Finnish land policy context, and each intervention mechanism is portrayed by two to six identified land policy indicators. In the following sections, we describe in more detail the intervention mechanisms, the related land policy indicators, and their scale of variation and distribution in the context of Finnish land policy. Tables 2–8 at the end of each section summarize the indicators in each intervention mechanism.<sup>9</sup> To preserve the chronological order of different interventions, we start the discussion with intervention mechanisms related to time-related indirect effects in e.g., the land acquisition and planning processes, followed by those related to direct effects on the (marginal) cost of development.

## 4.1. Indirect effects through time-related interventions in the development process

The first group of intervention mechanisms relates to less tangible impacts on marginal cost of development caused by development uncertainty and various delays. The delays may relate to different stages of the development process, such as planning (e.g., Ball, 2011), the permitting process (e.g., Mayer and Somerville, 2000) or the implementation of plans (e.g., Alexander, 1998). We identify four intervention mechanisms that relate to indirect impacts on marginal costs caused by process delays or uncertainty.

#### 4.1.1. Municipal land reserve building

Municipal land reserve building describes the land acquisition and land banking behaviour of municipalities. The possession of land, and subsequently the supply of buildable land, has a crucial role in facilitating housing development in the context of countries where municipalities are closely involved in land development and where a large proportion of housing development occurs on public (municipal) land. An adequate holding of public land for development purposes enables municipalities to respond more flexibly and timely to changes in local housing demand (Barlow, 1993), because of which we would in general also expect a more responsive housing supply. However, it should be noted that in other institutional contexts of land policy, where, for example, private land development is the dominant approach, municipal land reserve building may have much more limited influence on housing development. In such contexts, land reserve building may be done to a much larger extent by private actors for speculative purposes (see e.g., Triantafyllopoulos, 2017; Wang et al., 2019; Murray; 2020), although it can also provide the municipalities with a tool to, for instance, foster redevelopment (e.g., Tappendorf and Denzin, 2011; Alexander, 2008). In the Finnish context, three indicators measure municipalities' land reserve building.

4.1.1.1. Land acquisition activity and land holdings. Although public land acquisition and land ownership generally play a major role in the Finnish land market, local conditions do vary. The indicator captures two things: how actively and systematically a municipality acquires land from private landowners for public land banking purposes; and the

(+) Increasing effect

(-) Decreasing effect

(+/-) Increasing or decreasing effect depending on implementation of intervention.

<sup>&</sup>lt;sup>9</sup> To demonstrate more carefully how the land policy interventions work, we use the indicator *Reminder to build* as an illustrative example in Appendix B.

extent of existing land holdings. Typically, municipalities acquire socalled raw land that is mainly unbuilt but expected to be planned for development in the near future (e.g., Viitanen et al., 2003). It is to note, that the price paid in such transactions, both voluntary and compulsory, is set to reflect land values before planning. In 2020, there were in total 180 voluntary transactions of raw land in the market with an average area of 6,0 ha and an average price of  $1,63 \notin m^2$  (NLS, 2020). Due to the strong public planning control (see Section 3.2. in this paper), municipalities are nearly the only purchasers of raw land. The most common approach for the studied municipalities (70% of the sample) is to practice land acquisition very actively and systematically; if they don't, they already have extensive public land banks to secure development.<sup>10</sup> Only one municipality (3%) can be considered to display the other end of the scale, adopting a clearly inactive role on the land market yet having only limited public land ownership. The rest of the municipalities (27%) fall somewhere in-between: their land acquisition is systematic, but there is no direct pressure because they have adequate land banks.

4.1.1.2. Pre-emption of land. Pre-emption of land gives municipalities the right of first refusal when private property is sold, and it can provide them with an efficient tool for pursuing their land policy objectives (Nahrath, 2018). The indicator describes the prevalence and alignment towards the use of pre-emption of land. Half of the municipalities find the use of pre-emption viable even though they rarely use it, whereas it is considered viable and is fairly regularly used in 37% of the municipalities. The remaining municipalities (13%) use the instrument rarely or not at all.

4.1.1.3. Expropriation. The land policy indicator of Expropriation reflects the use and attitude towards the use of compulsory instruments that allow municipalities to expropriate land 'when the general need so demands' (Land Use and Building Act, 1999). Expropriation plays a different role in different municipalities: for most it is either not considered viable (53%) or considered viable yet very rarely used (40%), whereas in a few municipalities (7%) it is considered a standard instrument of land policy and used rather actively. Although the instrument is in general rarely used, the mere credible threat of its use may support municipalities in implementing their land development strategies, as it can also help to effectuate voluntary land sales (Buitelaar, 2010; van der Krabben and Jacobs, 2013).

#### 4.1.2. Delay in planning

Delay in planning describes different types of time-related (marginal) costs in the plan-making process that influence housing development. If the planning system is not working efficiently, the ability of new housing supply to respond effectively to changes in demand will be reduced (Ball, 2011; Mayer and Somerville, 2000). Here, delay in planning is characterized through five indicators.

4.1.2.1. Stakeholder involvement. The indicator Stakeholder involvement captures the degree of stakeholder involvement in the plan making process, which may emerge through a multi-staged decision-making process within the municipality or through structured deliberation, for example. A higher degree of involvement in the planning process, and thus a higher number of potential veto points in the process, can be understood to reflect a higher level of existing restrictiveness (Gyourko et al., 2021, 2008). Consequently, we would expect this to lower the responsiveness of housing supply. While most municipalities (57%) experience moderate stakeholder involvement in the planning process, there are a number of municipalities (27%) where several stakeholders

#### Table 2

Summary of land policy indicators related to municipal land reserve building and their variation.

| Indicator   | Indicator scale   | Share of municipalities |
|---|---|-------------------------|
| Land acquisition<br>activity and land<br>holdings | 1 = Land acquisition is not<br>systematic, and municipality has low<br>land ownership   | 3%                      |
| C C   | 2 = Land acquisition is systematic,<br>and municipality is a major<br>landowner (no direct pressure on<br>land acquisition)   | 27%                     |
|   | 3 = Land acquisition is considered<br>highly important, and it is done<br>actively and systematically, and/or<br>municipality has a high degree of<br>(relevant) land ownership | 70%                     |
| Pre-emption of land                               | 1 = Pre-emption is not used or very rarely used   | 13%                     |
|   | 2 = Pre-emption is rarely used but viable   | 50%                     |
|   | 3 = Pre-emption is used fairly<br>regularly when the municipality has<br>the option   | 37%                     |
| Expropriation                                     | 1 = Expropriation has not been used<br>or it is not viable  | 53%                     |
|   | 2 = Expropriation is rarely used but viable   | 40%                     |
|   | 3 = Expropriation is used rather actively   | 7%                      |

are involved and/or involvement in the process is substantial. Just less than one-fifth of the municipalities do not feel that stakeholder involvement notably hampers the planning process and/or feel that stakeholder involvement in the process is minor.

4.1.2.2. Stakeholder pressure. The implicit strategic behaviour of local authorities and stakeholders participating in the planning process may, however, also be of importance for the planning delay and thus also for the housing supply (in)responsiveness. This aspect of planning delay is captured by the indicator *Stakeholder pressure*. In some municipalities, local planning authorities may feel under pressure to step up housing development and growth (30%), which may be regarded as a positive trait from the perspective of supply responsiveness. However, a larger share of the municipalities (43%) experiences stakeholder pressure mainly geared to opposing growth, which is conversely expected to lower the responsiveness of housing supply.

4.1.2.3. Planning delay. The indicator captures the delay in the development process caused by the planning process. It describes the typical length of the detailed planning process in the urban area from initiation to approval, where a longer process is expected to cause a lower responsiveness of housing supply. The indicator combines the municipal authorities' estimated typical time for preparing new detailed plans and for making alterations to existing detailed plans, and it aims to reflect the average planning delay. Although the planning delay may vary systematically with plan size, there is clear variation between municipalities in *typical* planning times. These vary from less than a year (20%) to well over a year (47%).

4.1.2.4. Plan appeals. The indicator Plan appeals captures the experienced prevalence of appeals to local detailed plans in the municipality. As local detailed plans have statutory force, they are subject to rights of appeal by residents, organizations or authorities, which may further delay the planning process and thus lower the responsiveness of housing supply. While many of the municipalities (47%) do not experience considerable resistance towards planning decisions through appeals, more than one-fifth (23%) say that plans are often appealed.

<sup>&</sup>lt;sup>10</sup> It is noteworthy that historical factors play a significant role in the diversity of extant landownership patterns. Some municipalities have received substantial land allocations from the state in the past, which is still today reflected in high landownership shares in those municipalities.

4.1.2.5. (Weak) State of municipal finances. This indicator captures whether the municipality's financial status negatively affects their possibilities to carry out planning and development of residential areas and thus has an impact on planning delay and thereby on the housing supply (in)responsiveness. In Finland, municipalities are charged with providing statutory public services and service infrastructure, and therefore the state of municipal finances has an important role in securing land development. Across the municipalities, 50% experience that financial situation adversely impacts planning and housing development. The remaining municipalities do not experience any significant impact (17%) or face no impact at all (33%).

#### 4.1.3. Delay in availability of buildable land

Delay in availability of buildable land describes how efficiently the building plots are supplied for construction. The intervention mechanism follows the notion that releasing more building land increases housing supply (e.g., White and Allmendinger, 2003). This type of intervention mechanism is particularly imperative in the context of public land development, such as in Finland, where the public authority ultimately transfers serviced building plots for housing development (see e.g., Valtonen et al., 2017a). In other country contexts, delay in availability of buildable land may to a greater extent be subject to, for example, private land reserves and building plots that are withheld in

#### Table 3

Summary of land policy indicators related to delay in planning and their variation.

| Indicator   | Indicator scale   | Share of<br>municipalities |
|---|---|----------------------------|
| Stakeholder involvement   | 1 = Several stakeholders<br>participate actively in the planning<br>process, or involvement is<br>substantial       | 27%                        |
|   | 2 = Different stakeholders<br>participate moderately in the<br>planning process                                     | 57%                        |
|   | 3 = Stakeholder involvement is<br>minor; stakeholder involvement<br>does not notably hamper the<br>planning process | 17%                        |
| Stakeholder pressure  | 1 = Stakeholder pressure is mainly opposed to growth  | 43%                        |
|   | 2 = Stakeholder pressure is both<br>opposed to growth and supportive<br>of growth                                   | 27%                        |
|   | 3 = Stakeholder pressure is mainly supportive of growth   | 30%                        |
| Planning delay (incl. new<br>detailed plans and<br>alterations) | 1 = Planning process is typically<br>rather time consuming (more than<br>a year)                                    | 47%                        |
|   | 2 = Planning process typically<br>takes a moderate amount of time<br>(approx. 1 year)                               | 33%                        |
|   | 3 = Planning process is typically<br>relatively prompt (below 1 year)   | 20%                        |
| Plan appeals  | 1 = There are often appeals to the plans  | 23%                        |
|   | 2 = There are sometimes appeals to<br>the plans, affecting development<br>somewhat                                  | 30%                        |
|   | 3 = Appeals do not considerably<br>affect planning and development  | 47%                        |
| (Weak) State of<br>municipal finance                            | 1 = Financial status hinders the<br>planning of residential areas and/<br>or affects development                    | 50%                        |
|   | 2 = Financial status might have<br>some impact but does not<br>significantly hinder planning and/<br>or development | 17%                        |
|   | 3 =Financial status is adequate or<br>good and does not affect planning<br>and/or development                       | 33%                        |

anticipation of future gains (see e.g., Murray, 2020). In the Finnish context, the mechanism is featured through two indicators.

4.1.3.1. Conveyance process delay. The conveyance process delay is an inevitable part of the development process as public land development is a standard approach. The indicator describes the delay in building plot sales (or leases) by combining the experienced average delay of conveyance processes from initiation to agreement, thus also including extension periods to reservation times. While reservation times could be regarded to reflect more the delay in plan implementation rather than the delay in the availability of buildable land, it is included here for two reasons. First, many municipalities regard reservation times as part of the conveyance process delay, which complicates the task of extracting information on reservation times from the data. Second, final agreements of conveyance are in some cases only signed after approval of the detailed plan, making it a municipal decision. In general, any delays in the development process are expected to lower the responsiveness of housing supply. In almost half (47%) of the municipalities, the process typically takes less than a year and extensions are rarely granted, whereas in 10% of the municipalities the conveyance processes typically last more than a year and/or there are more flexible attitudes towards extensions of reservation times.

4.1.3.2. Planning of privately owned land for new residential areas. The indicator captures the municipality's strategic position towards the allocation of buildable plots on privately owned land. While the Finnish planning system is known for its widespread use of public land development, it provides for discretionary room through the planning and development of privately owned land and through the involvement of private property developers in the process (Valtonen et al., 2017b). While the landowners do have the right to initiate local detailed plans, it is ultimately at the municipalities discretion to decide whether a planning process is started or not. Also, while the time from start to completion is generally longer for local detailed plans on privately owned land than for plans on publicly owned land (Rinkinen and Kinnunen, 2017), the strategic decision to allocate buildable plots on privately owned land may restrain the total process delay - and thus increase the responsiveness of housing supply – as it is necessary neither to acquire nor to convey the land beforehand. Virtually all Finnish municipalities commonly carry out redevelopment and infill development of privately owned land, resulting in limited variation in this practice across municipalities. Therefore, the focus here is particularly on the position towards the allocation of plots on privately owned greenfield sites, which displays more variation across municipalities. The most common approach for municipalities is to plan new residential areas on publicly owned land only - 47% of the municipalities employ this approach. This suggests that the municipality often obtains ownership of the land or utilizes land from its landbanks before initiating the planning process. However, 17% of the municipalities allocate a large share of their buildable plots on privately owned land at the agreement with the landowners. Alternatively, the municipality has no restrictions on planning privately owned land as well.

#### 4.1.4. Delay in plan implementation

While the planning process can be seen as a form of 'frame setting' (Alexander, 1998, p. 304), the plan itself cannot secure any implementation. To facilitate the implementation of land use plans, municipalities may apply different types of land policy interventions. *Delay in plan implementation* in the Finnish context describes the time-related interventions influencing the actual realization of legally-binding local detailed plans, which can be materialized through the permitting and

#### Table 4

Summary of land policy indicators related to delay in the availability of buildable land and their variation.

| Indicator                              | Indicator scale  | Share of municipalities |
|--|--|-------------------------|
| Conveyance process<br>delay            | 1 = Average duration of conveyance<br>process is more than a year, and/or<br>extensions of reservations are readily<br>granted | 10%                     |
|  | 2 = Average duration of conveyance<br>process is approx. one year; extensions<br>of reservations are sometimes granted         | 43%                     |
|  | 3 = Average duration of conveyance<br>process is less than a year, and/or<br>extensions of reservations are rarely<br>granted  | 47%                     |
| Planning of<br>privately owned<br>land | 1 = Only on publicly owned land, in<br>some special cases on privately owned<br>land   | 47%                     |
| for new<br>residential areas           | 2 = Mainly on publicly owned land, occasionally on privately owned land  | 37%                     |
|  | 3 = Mainly on privately owned land; no<br>negative stance towards planning<br>privately owned land                             | 17%                     |

development control processes.<sup>11</sup> Here, the mechanism is featured through three indicators.

4.1.4.1. Post-planning process. The indicator describing the postplanning process summarizes the typical delay time in cadastral procedures. This includes activities such as plot parcelling and plot subdivisions according to local detailed plans, of which plot subdivisions appeared to be the most significant factor influencing the variation in process duration. A building permit can only be granted to a registered plot with a plot subdivision, and a delay in this stage of the process is therefore expected to lower the responsiveness of housing supply. In many municipalities (53%) the plot subdivision can be approved as part of the local detailed plan, which can compress the post-planning process to less than a month on average. In more than one-fifth (23%) of the municipalities, however, the plot subdivision is typically done as a separate process, which may suggest a delay time of several months in post-planning processes.

4.1.4.2. Conveyance stipulation of building obligation. The indicator measures the applicability and strictness of building obligations in the municipality's conveyance stipulations as well as the prevalence of extensions to the obligations. In general, we would expect a stricter conveyance stipulation to yield in a higher housing supply responsiveness. The majority of municipalities (70%) will in certain situations grant extensions to building obligations for legitimate reasons, whereas 17% of the municipalities employ obligations to which extensions are readily granted – indicating rather flexible processes with likely delays. The obligations regime is stricter in 13% of the municipalities: in these cases, extensions are granted only judiciously.

4.1.4.3. Reminder to build. The indicator Reminder to build summarizes the municipality's alignment towards the use of the corresponding instrument. It entitles the municipality to issue an owner or titleholder of a plot with a reminder to build if less than half of the gross floor area permitted for the plot has been used. If the building plot is not developed within three years, the municipality is entitled to expropriate it without special permission. Municipalities can use a reminder to build to

facilitate housing development and thus to contribute to a higher responsiveness of housing supply. This tool is systematically used in 10% of the municipalities, whereas the majority (70%) find the instrument rather ineffective for securing coordinated development and hence do not currently use it.

#### 4.2. Direct effects on the development process

The second avenue of intervention mechanisms are interventions with direct effects, which we define as ones that directly influence the cost of construction. Direct cost effects can arise, among other things, through fees or restrictions on land use. These cost-effects can make new development or redevelopment unfeasible in areas where market values are below construction costs (Gyourko and Saiz, 2004) and thus lower the responsiveness of housing supply. While in certain scenarios the cost burden can be shifted to landowners - thereby driving down the price developers are willing to pay for land - it would ultimately lead to a decrease in land supply. This could further diminish the responsiveness of housing supply. Following this notion, we will not differentiate between these two channels in the subsequent sections. However, it should be noted that local authorities can also encourage housing development through financial incentives, such as reductions or allowances, which may increase project viability and thus make development attainable or larger scale. We identify three intervention mechanisms that relate to direct construction cost-impacts caused by land policy interventions.

#### 4.2.1. Production costs and impacts on profit margin

*Production costs and impacts on profit margin* describes the use of land policy interventions that may impact the cost of development and profit margins of development projects. Taking the stance of economic theory, housing supply will be increased if the (marginal) cost of development declines as a result of land policy interventions (McFarlane et al., 2021). Equally, such interventions that increase the (marginal) development costs will influence the supply of housing negatively. Here, we have identified six indicators.

4.2.1.1. Land value capture practices. The land value capture is a tool with which the municipalities may reap some of the increment value attributable to planning decisions (Alterman, 2012). When privately owned land is (re)developed, the municipality and private landowner

#### Table 5

Summary of land policy indicators related to delay in plan implementation and their variation.

| Indicator                    | Indicator scale   | Share of municipalities |
|------------------------------|---|-------------------------|
| Post-planning process        | 1 = Post-planning process is typically<br>rather time consuming (more than 2<br>months)         | 23%                     |
|                              | 2 = Post-planning process typically<br>takes a moderate amount of time<br>(approx. 1–2 months)  | 23%                     |
|                              | 3 = Post-planning process is typically relatively prompt (less than 1 month)                    | 53%                     |
| Conveyance<br>stipulation of | 1 = Municipality has an obligation to which extensions are readily granted                      | 17%                     |
| building obligation          | 2 = Municipality has an obligation to<br>which extensions are granted for<br>legitimate reasons | 70%                     |
|                              | 3 – Municipality has a strict<br>obligation regime, extensions are<br>only judiciously granted  | 13%                     |
| Reminder to build            | 1 = Reminders to build are not<br>currently used  | 70%                     |
|                              | 2 = Reminders to build have been used, expropriation has not been used                          | 20%                     |
|                              | 3 = Reminders to build have been used, expropriation has been used                              | 10%                     |

<sup>&</sup>lt;sup>11</sup> Ultimately, the implementation of land use plans is also dependent on decisions made by developers, for instance. Public regulatory interventions intended to impact the decision-making of others than the municipality itself are also included in mechanism VII (*Builders' cost of option to delay*).

agree upon their respective rights and obligations related to the development by signing a land use agreement. Municipalities have the discretion to decide on the terms and conditions of land use agreements and, for example, on how to define the threshold level of considerable financial gain for the property owner, below which land use fees are not imposed. The indicator Land value capture practices describes how a municipality determines the land development fee in land use agreements in comparison to the compulsory fee, where a lower fee is expected to reduce the marginal cost of development and thus increase the responsiveness of housing supply.<sup>12</sup> More specifically, the indicator describes how municipalities have set their threshold levels of considerable financial gain for the property owner, below which land use fees are not imposed. Almost one-third (30%) of the municipalities have no threshold levels for considerable financial gain, or the level is unfavourable to the landowner compared to a compulsory land development fee. However, most municipalities (57%) apply a similar threshold as defined for the compulsory land development fee. A small proportion of municipalities (13%) have threshold levels that are higher compared to the compulsory land development fee and can thus be viewed as providing an incentive to the landowner.

4.2.1.2. Incentives in land use agreements. To facilitate redevelopment on privately owned plots, municipalities can at their own discretion decide to offer economic incentives in land use agreements, such as discounts on the land development fee when development takes place on a plot owned by a residential housing company. Such incentives are expected to lower the marginal cost of development and thus increase the housing supply responsiveness. 37% of the municipalities have no clear incentives to support the development of privately owned land, while some municipalities (17%) apply several or particularly favourable incentives in order to increase the supply of new housing. The remaining municipalities (47%) have some incentives to support housing development on privately owned land.

4.2.1.3. Inclusionary housing requirements. The indicator of Inclusionary housing requirements describes how widely the municipality uses developer obligations on affordable housing, such as different quotas steering the provision of affordable housing, which are expected to increase the marginal cost of development and thus lower the housing supply responsiveness. Some municipalities (27%) actively use inclusionary housing requirements, while many (33%) do not generally apply such requirements. The remaining municipalities (40%) may have some areaspecific requirements or apply them moderately.

4.2.1.4. Residential structure requirements. The indicator Residential structure requirements captures the prevalence of constraints on developers' decisions related to housing structure, such as family apartment requirements or minimum area requirements on housing units. The application of binding requirements is expected to lower the responsiveness of housing supply. 23% of the municipalities actively use residential structure requirements, while 43% do not generally apply such requirements. The remaining 33% may use some area-specific requirements or apply them moderately.

4.2.1.5. Requirements in detailed plans. The indicator depicts the pervasiveness of planning-related requirements on housing, such as density, design and quality ordinances, which may impact development costs through the need for more expensive building materials or loss of flexibility in product design, for example. Close to one-third or 30% of

the municipalities have rather detailed plans, while 13% can be considered to have rather flexible plans. The remaining 57% have mixed requirements with flexible plans in some areas and stricter plans in others.

4.2.1.6. Objects of protection. The indicator Objects of protection summarizes the prevalence of local interventions that aim to protect environmental sites, landscapes or buildings. While such interventions may be needed from, for example, an environmental perspective, they may induce costs or restrain the housing development potential. 27% of the municipalities can be considered to have an abundance of designations of protection, which hinders housing development, whereas 13% do not experience such designations to affect development.

#### 4.2.2. Market competition

*Market competition* relates to competition in land and housing development markets and the ways through which the municipality may interfere in that competition by the means of land policy interventions. This intervention mechanism stems from the notion that regulation and municipal decision-making which harm competition in the market, and thus increase the market power of some actors, may impact the costs and output of new housing (Coiacetto, 2009). Depending on the institutional

#### Table 6

Summary of land policy indicators related to cost of development and impacts on profit margin and their variation.

| Indicator                               | Indicator scale   | Share of municipalities |
|---|---|-------------------------|
| Land value capturing<br>practices       | 1 = No threshold in development fee<br>or unfavourable to landowner   | 30%                     |
| -                                       | 2 = Threshold set at medium range<br>(similar to development compensation<br>500 sqm)                               | 57%                     |
|   | 3 = Threshold favourable to landowner   | 13%                     |
| Incentives in land use<br>agreements    | 1 = No clear incentives supporting development of privately owned land  | 37%                     |
| C C                                     | 2 = Some incentives supporting<br>development of privately owned land   | 47%                     |
|   | 3 = Multiple/particularly favourable incentives   | 17%                     |
| Inclusionary housing<br>requirements    | 1 = Municipality actively uses<br>requirements on affordable housing  | 27%                     |
| requirementerite                        | 2 = Municipality has some area-<br>specific requirements/uses<br>requirements (moderately)                          | 40%                     |
|   | 3 = Municipality has no general requirements on affordable housing  | 33%                     |
| Residential structure requirements      | 1 = Municipality actively uses<br>requirements on residential structure   | 23%                     |
| 1                                       | 2 = Municipality has some area-<br>specific requirements/uses<br>(moderately)                                       | 33%                     |
|   | 3 = Municipality has no general requirements on residential structure   | 43%                     |
| Requirements of local<br>detailed plans | 1 = The municipality has quite detailed plans overall   | 30%                     |
| F                                       | 2 = The municipality might have<br>rather detailed plans in some areas but<br>more flexible in others               | 57%                     |
|   | 3 = The municipality has relatively flexible plans overall  | 13%                     |
| Objects of protection                   | 1 = The municipality has a lot of<br>objects of protection, which hinders<br>development and growth                 | 27%                     |
|   | 2 = The municipality has some objects<br>of protection and they might<br>sometimes hinder development and<br>growth | 60%                     |
|   | 3 = The municipality has few objects<br>of protection and/or they do not affect<br>development and growth           | 13%                     |

 $<sup>^{12}</sup>$  A municipality may collect a compulsory land development fee if an agreement is not reached with the landowner. The maximum fee is 60% of the increase in the plot value. If the amount of building rights or increase in building rights does not exceed 500 m2 in gross floor area, the landowner cannot be required to pay a fee.

arrangements in place, such interventions can occur in several different stages of the development process. In the Finnish context, the mechanism is described through two indicators.

4.2.2.1. Developer steering. The indicator Developer steering describes the ways in which the municipalities may intervene in market competition – whether they actively advocate competition among developers on the market by actively advertising development opportunities, for example, or whether they constrain competition by imposing developerspecific development caps or other artificial measures. The latter is the custom in 20% of the municipalities, where the most common way is to restrict the number of parallel development projects. Competition is actively advocated in 7% of the municipalities.

4.2.2.2. Land market constraints. Land market constraints relate to public interventions on the land market that may limit the functions of private actors on the market and thus serve as anti-competitive practices. Ultimately, such practices can reduce the responsiveness of housing supply. It aims more specifically to capture whether the municipality has adopted any restrictive policies on private actors in the land market by refusing to convey buildable plots, for example. In Finland, the most common type of such restrictions is targeted towards land investment funds, who acquire land and lease it to (typically) multifamily housing projects. Some municipalities do not convey buildable plots to these funds at all (17%), while others convey rarely and judiciously (47%). Several municipalities (37%) have no restrictions towards land investment funds.

#### 4.2.3. Builders' cost of option to delay

Builder's cost of option to delay describes the presence and use of land policy instruments and practices that financially incentivize development activity by increasing the cost of option to delay and thus the responsiveness of housing supply. In contrast to the intervention mechanism measuring delays in plan implementation, builders' cost of option to delay measures the enforcement of direct negative cost impacts if development is delayed by the choice of the landowner or developer. The mechanism is featured through two indicators.

4.2.3.1. Applicability of contractual penalties. The indicator portrays the monitoring of the fulfilment of contractual building obligations and the applicability of economic sanctions in case of non-compliance. While some municipalities do not enforce economic sanctions in practice or charge them only rarely (17%), others are more active in both monitoring and charging sanctions (37%).

#### Table 7

Summary of land policy indicators related to market competition and their variation.

| Indicator               | Indicator scale   | Share of municipalities |
|-------------------------|---|-------------------------|
| Developer<br>steering   | 1 = Municipality has restricted the number<br>of developer projects or clearly limits<br>competition                                | 20%                     |
|                         | 2 = Municipality may have some ways of<br>promoting developers/companies on the<br>market or has a neutral stance to<br>competition | 73%                     |
|                         | 3 = Municipality actively promotes/<br>contacts developers  | 7%                      |
| Land market constraints | 1 = Municipality does not convey to property funds  | 17%                     |
|                         | 2 = Municipality rarely/judiciously<br>conveys to property funds/property funds<br>not on the market                                | 47%                     |
|                         | 3 = Municipality conveys to property<br>funds/has no negative position towards<br>land property funds                               | 37%                     |

4.2.3.2. Vacant urban land tax. The indicator describes the use of vacant urban land tax in municipalities. A vacant urban land tax allows municipalities to apply a higher tax rate (2%-6%) on undeveloped buildable plots, and it may thus be used to encourage housing construction. The current Finnish property tax system obliges certain municipalities in the Helsinki region to apply a vacant urban land tax of at least 3 percentage points above the general real estate tax rate, but in all other municipalities the applicability and chosen level of a vacant urban land tax is optional. Taking these legislative directives into account, the results show that 17% of the municipalities have not applied any vacant urban tax, while the same proportion (17%) has a tax set at or close to the maximum range.

#### 5. Discussion

Despite the rapid growth of research on different types of land policy interventions, particularly land use regulations, and their impacts on housing development and housing supply (e.g., Gyourko and Molloy, 2015; Quigley and Rosenthal, 2005; White and Allmendinger, 2003), the complexities of this relationship still remain inadequately understood. To bring clarity to this relationship and to help disentangle the many ways in which local public authorities may and *de facto* intervene in land markets and housing construction, this paper set out to conceptualize intervention mechanisms. Essentially, these intervention mechanisms distinguish seven paths through which local land policy interventions may influence the (in)responsiveness – and thereby the quantity – of housing supply.

Methodologically our study combines prior theoretical knowledge, particularly from urban economics literature, with empirical data from the Finnish land policy context. Therefore, while land policy interventions vary between different institutional regimes and operational contexts, the proposed intervention mechanisms should not be very context-specific considering that they are grounded in economic theory and the idea of two basic impact channels (i.e., direct effects and indirect effects). We note, however, that in different environments the degree of prevalence of different intervention mechanisms may vary substantially. For example, in country contexts where the local public authority does not control the supply of buildable land to the same extent as many Finnish municipalities do, the intervention mechanism called Market competition might have less importance from the perspective of (in) responsiveness of housing supply. Therefore, some adjustment is necessary when applying the approach in other country contexts, particularly regarding the indicators as they stem from the empirical context.

In line with previous studies on the use of different land policy instruments (e.g., Krigsholm et al., 2022), we observe both variation and uniformity in the ways the instruments are applied across municipalities. Our findings imply, for example, that Finnish municipalities do not

Table 8

Summary of land policy indicators related to builder's cost of option to delay and their variation.

| Indicator                                    | Indicator scale   | Share of municipalities |
|--|---|-------------------------|
| Applicability of<br>contractual<br>penalties | 1 = Obligations sporadically<br>monitored, no economic sanctions<br>charged or rarely charged | 17%                     |
|  | 2 = Obligations monitored; economic sanctions rarely charged                                  | 47%                     |
|  | 3 = Obligations monitored actively;<br>economic sanctions charged                             | 37%                     |
| Vacant urban land<br>tax                     | 1 = Municipality does not have a tax<br>on unbuilt sites                                      | 17%                     |
|  | 2 = Municipality has tax set at low to<br>"medium range"                                      | 67%                     |
|  | 3 = Municipality has tax set at or close<br>to maximum  | 17%                     |

differ very much in their practices concerning contractual building obligations in public land allocations, whereas we observe much more variation in some other contractual stipulations between a municipality and a developer and/or a landowner, such as residential structure requirements. Interestingly, we observe that some municipalities have established practices that promote competition in the market for housing development, whereas others may (unintentionally) restrain competition.

Although the significance of different interventions on the supply (in)responsiveness is an empirical question, the variation in the deployment of 'housing supply relevant' land policy interventions suggests that their cumulative influence on the responsiveness of housing supply also varies at the municipal level. Earlier studies quantifying the extent and strength of local land use regulation, particularly in the context of the US and UK, have made similar observations (e.g., Ball, 2011; Bramley, 1998; Bramley and Watkins, 2014; Gyourko et al., 2021, 2008; Jackson, 2016, 2018).

To illustrate this variation in how the different choices of municipalities can pile up into different degrees of influence on the housing supply responsiveness, there is, however, a need to move beyond purely descriptive analysis of the ways municipalities apply housing supply relevant instruments and practices. One option is to rate the municipalities in a way that describes their relative degree of *housing supply restrictiveness*. In this context, housing supply restrictiveness refers to the extent or degree to which municipalities apply housing supply-relevant land policy interventions, influencing the responsiveness of housing supply. Therefore, it reflects how the implementation of these interventions may create conditions that either facilitate or impede the effective and timely provision of housing within a municipality. For this illustrative purpose, we can make use of the scaling schemes and the combination of individual scores assigned to each municipality.

To achieve this, we summarize the land policy indicator scores for each municipality, where a lower (higher) value reflects a relatively more (less) housing supply restrictive land policy regime.<sup>13</sup> Although the scaling scheme presented in this paper indicates only an ordered structure and not numerical values in a mathematical sense, it offers a way to group the municipalities and give insight into the structures of the underlying data. Therefore, we use these aggregate scores to categorize the municipalities into three quantiles (i.e., terciles) describing their relative intensity of applying interventions that may restrict housing supply (see Table 9). Here, the most restrictive municipalities account for the tercile with the lowest aggregate scores, and the least restrictive municipalities account for the tercile with the highest aggregate scores. Consequently, the mid-tercile account for a moderate level of restrictiveness.

The categorization can further be utilized to examine how municipalities differ in terms of the relative restrictiveness across intervention mechanisms. By summarizing the scores of the land policy indicators pertaining to each intervention mechanism, we can in a similar vein categorize the municipalities into terciles that describe the relative restrictiveness across different intervention mechanisms (see the checkmark indications in Table 9). These categorizations demonstrate that municipalities that intervene in a relatively more restrictive way overall do so generally through both indirect and direct intervention mechanisms. The same applies to the other end of the spectrum: less restrictive municipalities also tend to be more generous in applying interventions that incentivize housing supply. While there are no evident patterns describing these relations, the illustration suggests that there is an

apparent risk that the choices of land policy interventions may (unknowingly) pile up into a more housing supply restrictive environment in some municipalities compared with others. We note, however, that the variations observed in the application of different land policy interventions across municipalities and, consequently, the differences in the degree of housing supply restrictiveness, may be influenced by differences in local market environments rather than solely by the deliberate choices made by policymakers. For instance, municipalities facing a declining population trend and lower demand might have different opportunities to employ distinct interventions compared to those municipalities with increasing population trend and high demand. This suggests the need to consider also external factors when examining and evaluating how municipalities can alleviate the housing supply restrictiveness.

Moreover, it should be noted that this study does not intend to develop exact measures of housing supply restrictiveness of local land policy but rather showcase an approach that can be used to identify and compare the supply restrictiveness of different land policy decisionmaking processes and practices. Therefore, weightings of different intervention mechanisms and land policy indicators are not used. This is done for simplicity, although we acknowledge that the different indicators and intervention mechanisms are unlikely to have equal impacts on housing supply (see e.g., Cheshire and Vermeulen, 2009; Gyourko and Molloy, 2015). For example, land policy indicators with either positive or negative influence on housing supply could be of different importance for the overall restrictiveness. Land policy indicators associated with different intervention mechanisms can also be interrelated. This suggests that land policy interventions incurring greater influence in the early stages of the housing development process might affect the outcomes of interventions in later stages of the process (Mettepenningen et al., 2011). It remains a question for future research to quantify empirically reliable measures of housing supply restrictiveness of local land policy and examine their impacts on housing supply outcomes.

The construction of measures aimed at summarizing complex concepts, such as the housing supply restrictiveness of local land policy interventions, is always a subjective process shaped by the quality of the underlying data (OECD, 2008). In contrast to many extant studies using mainly questionnaires and official documents (see e.g., Ball, 2011; Bramley, 1998; Bramley and Watkins, 2014; Gyourko et al., 2008, 2021), this study set out to examine the variation in land policy practices primarily using interview data. Recently, some scholars have made the criticism that objective questions about local land policy interventions in questionnaires might yield strikingly different responses even in the same municipalities when asked just months apart (Lewis and Marantz, 2019). Others have responded that careful questionnaire design can mitigate such issues (Pendall, 2020). While the use of interview data presents its own challenges, such as concerns of errors arising from individual biases or heuristics (Maestas et al., 2014), it does offer some clear benefits too. The use of interviews provides access to more in-depth information and tacit knowledge about land policy decision-making and current practices that surveys may not effectively address (Robson, 2002; Lewis and Marantz, 2019). Interviews may also reveal land use enactments that set permissible regimes and thresholds in municipal land policy agendas and other council-approved documents that are not binding in practice (e.g., Quigley and Rosenthal, 2005). Moreover, while our interview data is based on subjective interpretations by municipal land policy officials, such responses may help to characterize important elements of the local decision-making environment (Lewis and Marantz, 2019).

Finally, while we believe that it is of importance to advance our understanding of the influence of land policy interventions on the quantity of housing supply, it is important to acknowledge that it is not sufficient to judge the overall quality of different land policy practices and decision-making processes based on this view only. For example, a land policy intervention may considerably constrain the quantity of

<sup>&</sup>lt;sup>13</sup> Here, the indicators have been scored in such a way that the most disadvantageous or least beneficial mode of intervention from the perspective of supply responsiveness receives the lowest score, whereas the least disadvantageous and most beneficial receives the highest score.

#### Table 9

Illustration of the relative housing supply restrictiveness of local land policy in Finnish municipalities.

| Category          | Municipality | Population | Indirect intervention<br>mechanisms |     |            | Direct intervention mechanisms |     |     |          |
|-------------------|--------------|------------|-------------------------------------|-----|------------|--------------------------------|-----|-----|----------|
|                   |              |            | I.                                  | II. | III.       | IV.                            | v.  | VI. | VII.     |
| Most restrictive  | Vantaa       | 237,200    | <b>\</b> \\                         | 1   | 1          | 11                             | 1   | 11  | 1        |
| (✔)               | Joensuu      | 76,900     | 1                                   | 11  | 11         | 11                             | 1   | 1   | 1        |
|                   | Hämeenlinna  | 67,800     | 1                                   | 1   | <i>」」」</i> | 11                             | 1   | 111 | 1        |
|                   | Helsinki     | 656,900    | 11                                  | 11  | 1          | 11                             | 1   | 111 | <i>」</i> |
|                   | Espoo        | 292,800    | 1                                   | 11  | 111        | 1                              | 1   | 111 | 11       |
|                   | Tampere      | 241,000    | <i>」</i>                            | 1   | 111        | 1                              | 1   | 111 | 11       |
|                   | Jyväskylä    | 143,400    | <i>」</i>                            | 11  | 111        | 11                             | 1   | 1   | 1        |
|                   | Vaasa        | 67,600     | <i>」</i>                            | 1   | 1          | <i>」」」</i>                     | 11  | 11  | 11       |
|                   | Kirkkonummi  | 40,100     | 1                                   | 11  | 111        | 1                              | 1   | 11  | 11       |
|                   | Kerava       | 37,100     | 1                                   | 11  | 11         | 11                             | 1   | 1   | <i>」</i> |
|                   | Turku        | 194,400    | <i>」 」 」 」</i>                      | 1   | 111        | 1                              | 1   | 111 | 11       |
|                   | Kokkola      | 47,800     | 11                                  | 11  | 11         | 11                             | 111 | 1   | 11       |
| Moderate          | Salo         | 51,600     | 11                                  | 11  | 11         | 1                              | 111 | 11  | 11       |
| (√√)              | Lohja        | 45,900     | 111                                 | 1   | 1          | 111                            | 11  | 11  | 11       |
|                   | Nurmijärvi   | 43,700     | 111                                 | 11  | 111        | 1                              | 1   | 111 | 11       |
|                   | Pori         | 83,700     | 111                                 | 11  | 11         | 11                             | 111 | 1   | 11       |
|                   | Lappeenranta | 72,700     | 111                                 | 11  | 1          | 111                            | 11  | 1   | 11       |
|                   | Kotka        | 51,700     | 1                                   | 111 | 11         | 11                             | 111 | 11  | 1        |
|                   | Kuopio       | 120,200    | 111                                 | 11  | 111        | 111                            | 11  | 1   | 11       |
|                   | Seinäjoki    | 64,100     | 111                                 | 111 | 11         | 11                             | 11  | 1   | 11       |
|                   | Rovaniemi    | 63,500     | 1                                   | 111 | 111        | 111                            | 11  | 1   | 11       |
|                   | Rauma        | 39,000     | 11                                  | 11  | 1          | 11                             | 111 | 11  | 11       |
|                   | Tuusula      | 38,800     | 111                                 | 11  | 111        | 11                             | 111 | 111 | 11       |
| Least restrictive | Oulu         | 207,300    | 111                                 | 11  | 11         | 111                            | 111 | 1   | 1        |
| (√√√)             | Kouvola      | 81,200     | 1                                   | 111 | 1          | 11                             | 111 | 111 | 1        |
|                   | Porvoo       | 50,600     | 111                                 | 11  | 11         | 11                             | 111 | 111 | 11       |
|                   | Hyvinkää     | 46,600     | <i>」」」</i>                          | 111 | 1          | 11                             | 11  | 111 | 11       |
|                   | Kajaani      | 36,600     | 11                                  | 11  | 11         | 11                             | 111 | 11  | 11       |
|                   | Lahti        | 120,00     | 11                                  | 11  | 111        | 11                             | 111 | 111 | 11       |
|                   | Järvenpää    | 44,500     | 111                                 | 11  | 1          | 111                            | 111 | 111 | 111      |

Notes: The key  $\checkmark$  denotes most restrictive and/or least beneficial,  $\checkmark \checkmark$  denotes moderately restrictive and/or moderately beneficial, and  $\checkmark \checkmark \checkmark$  denotes least restrictive and/or most beneficial tercile of the municipalities. The categorization is based on the individual indicator scores assigned to each municipality, where the overall restrictiveness is assessed based on the total of all land policy indicator-scores. The mechanisms-specific rating is based on the sum of the indicator scores pertaining to a mechanism. The intervention mechanisms refer to I. Municipal land reserve building, II. Delay in planning, III. Delay in availability of buildable land, IV. Delay in plan implementation, V. Production costs and impact on profit margin, VI. Market competition and VII. Builders' cost of option to delay. The population represents the total population in the municipality in 2020, rounded to the nearest 100.

housing supply, but still be necessary to advance other policy goals, such as environmental sustainability (e.g., Bovet et al., 2018), social inclusion (e.g., Nelson et al., 2004; Bento et al., 2009), or the quality of housing stock (e.g., May, 2003; Wang, 2014). Taking into account the influence of land policy interventions on quantity of housing supply can, however, improve the comparative analysis of different practices. As municipalities can deliberately make choices of which interventions to apply to advance their public planning (and land policy) objectives, our findings suggest that by paying attention to the land policy practices that contribute to housing supply restrictiveness, municipalities may be able to identify such processes that secure their policy goals but minimize the negative subsidiary effects on the responsiveness of housing supply. The results can thus help policymakers to realize the influence of land policy interventions on the quantity of housing supply and shed light over the trade-offs of different interventions. Ultimately, this may promote the design and implementation of (better) land policies.

#### 6. Conclusions

This paper set out to address two main objectives. The first objective was to contribute to a better understanding of how local authorities may influence the (in)responsiveness of housing supply and, ultimately, the quantity of housing supply through land policy interventions. To that end we drew on the extant literature and empirical data in the Finnish land policy context to distinguish different conceptual categories of land policy interventions contributing to the (in)responsiveness of housing supply. We formulated seven *intervention mechanisms*: 1) Municipal land reserve building, 2) Delay in planning, 3) Delay in availability of

buildable land, 4) Delay in plan implementation, 5) Production costs and impact on profit margin, 6) Market competition and 7) Builders' cost of option to delay. The intervention mechanisms provide a structured view of the linkages between land policy interventions and the (in)responsiveness of housing supply and help to disentangle the myriad ways in which these interventions affect housing markets.

Second, this paper sought to use the intervention mechanisms to examine the variation in local land policy interventions that can be associated with housing supply restrictiveness. Our findings demonstrate that there are notable heterogeneities in the application of different interventions at the local level. While these findings showcase the variation in the application of individual indicators, the diversity may also have implications for the local housing supply restrictiveness. Moreover, these findings suggest that by paying attention to the land policy practices that contribute to the (in)responsiveness of housing supply, municipalities may alleviate negative subsidiary effects on the quantity of (market-rate) housing supply.

Our paper contributes to the literature on the influences of land policy interventions on housing markets. The intervention mechanisms formulated provide a higher degree of analytical clarity on the linkages between land policy interventions and housing supply, whereas the variation identified in the application of different interventions provides a structured way to examine the variation in the degree to which municipalities' land policies may constrain the housing supply responsiveness. Our findings can also provide certain benefits for municipal land policy experts and politicians in acknowledging the subsidiary effects that such interventions may cause and reflecting on their trade-offs. This may help them move towards more welfare-increasing outcomes. The results of this study suggest many avenues for further research. Potential topics include empirical analysis of the relative importance of the intervention mechanisms and the further development of housing supply restrictiveness measures at the municipal level. Not only can restrictiveness measures be used to identify the most and least restrictive municipalities, but they also offer plenty of scope for identifying the drivers behind the variation between these municipalities. Such measures can further be used to analyse the economics of land policy interventions, for example housing affordability, and to gain a better understanding of how to manage urban growth through land policies. Furthermore, the findings of this study serve as an opening to make assessments and comparisons of the implications and roles of the intervention mechanisms in different institutional regimes and land policy contexts.

#### Funding

This research was funded by the Academy of Finland (grant no. 327800, Smart Land project and grant no. 324393, Outcomes of land policies in real estate markets) and Government's Analysis, Assessment, and Research Activities (grant no. VN/14565/2019).

#### **Declaration of Competing Interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Data Availability

The authors do not have permission to share data.

#### Acknowledgements

The authors would like to thank Tuulia Puustinen for her help with data collection, as well as the participants of Research Seminar in Real Estate held at Aalto University in 2022 for their helpful comments on an earlier version of the paper. In addition, the authors would like to thank Oskari Harjunen for his helpful comments and suggestions on the paper. Working versions of the paper were presented at the ERES conference (online) in 2021 and the PLPR conference in Ghent 2022. The authors would like to thank for all the comments received there.

#### Appendix A. General list of identified land policy instruments and practices in the Finnish context

| Category          | Land policy instrument or practice                            | Final selection <sup>a</sup> |
|-------------------|---|------------------------------|
| Land policy       | Main goal or objective for land policy                        |                              |
|                   | Degree of land holdings                                       | х                            |
|                   | Land acquisition activity                                     | х                            |
|                   | Voluntary land acquisition                                    |                              |
|                   | Pre-emption of land   | х                            |
|                   | Expropriation   | х                            |
|                   | Pricing of undeveloped land                                   |                              |
|                   | Planning of privately owned land for new residential areas    | х                            |
|                   | Planning of privately owned land                              |                              |
|                   | through alterations of plans                                  |                              |
|                   | Use of land use agreements                                    |                              |
|                   | Land use development compensation                             |                              |
|                   | Land value capture practices                                  | х                            |
|                   | Incentives in land use agreements                             | x                            |
|                   | Compulsory land development fee                               |                              |
|                   | Special development areas                                     |                              |
|                   | Conveyance practices of building plots                        |                              |
|                   | Lease or sell decisions of building plots                     |                              |
|                   | Conveyance process delay                                      | х                            |
|                   | Conveyance decisions  |                              |
|                   | Reservation of sites  | х                            |
|                   | Conveyance stipulation of building obligation                 | x                            |
|                   | Applicability of contractual penalties                        | X                            |
|                   | Pricing of residential building right                         | A                            |
|                   | Follow up on building permits                                 |                              |
|                   | Stipulations of land use agreements                           |                              |
|                   | Vacant urban land tax   | х                            |
|                   | Reminder to build   | X                            |
|                   | Land market constraints                                       | x                            |
|                   | Developer steering  | x                            |
|                   | Inclusionary housing requirements                             |                              |
|                   | Practices of curbing housing price increases                  | х                            |
|                   | 0 01  | v                            |
| land was planning | Residential structure requirements<br>Stakeholder involvement | x                            |
| Land use planning |   | X                            |
|                   | Stakeholder pressure  | X                            |
|                   | State of municipal finance                                    | х                            |
|                   | Requirements of local detailed plans                          | х                            |
|                   | Alterations of plans  |                              |
|                   | Use of exceptional permits                                    |                              |
|                   | Objects of protection   | х                            |
|                   | Planning delay (incl. new plans and alterations)              | х                            |
|                   | Plan appeals  | х                            |
|                   | Post-planning process   | х                            |
|                   | Approval of building of dispersed settlements                 |                              |

<sup>a</sup>Some land policy instruments and practices were merged under one land policy indicator to reflect the data and the variation across municipalities in the best way possible

#### Appendix B. Finnish institutional framework: Reminder to build as an illustrative example of a land policy indicator

In Finland, the creating, updating, and approving of plans that allow building development is under the jurisdiction of municipalities. The municipalities have a so-called planning monopoly, which means that they can also independently decide when and where planning is started. This suggests that while the landowners or titleholders of a plot have the discretion to initiate land use plans, they have no real instrument that can secure the development of their land in urban areas. Moreover, the development of a property requires a building permit issued by the municipality, and in general, permits can only be granted if there exists a legally binding plan that allows the proposed development.

When a local detailed plan exists, the landowner is not in general obliged to build its plot (unless otherwise specified in e.g., conveyance contracts). To advance the implementation of the local detailed plan, the municipality has the right to issue an owner with a reminder to build when the plan has been in force for at least two years (Land Use and Building Act, 1999/32 §97). A reminder to build can only be issued if less than half of the allowed gross area is used, or the plot has been developed against the actual detailed plan. If the plot has not been built within three years after a reminder to build has been issued, the municipality is entitled to expropriate the plot without special permission. If the municipality expropriates the property, the landowner will receive full compensation (see e.g., Viitanen et al., 2003, p.61–62).

A reminder to build can be used to facilitate development and as a tool to reduce the landowner's possibility to withhold building land from the market. From a housing supply perspective, the reminder to build affects the delays in the development process. It aims to overcome the passivity of plan implementation by setting a time limit for the landowner to develop its plot. A reminder to build can therefore be understood as a similar public intervention as, for example, the building obligation that can be applied in the course of action of local land policy (see e.g., Hengstermann, 2018). By issuing reminders to build, the municipality may alleviate the delays of plan implementation and decrease the option value to wait to develop, and thus ultimately affect the time-related indirect costs. Moreover, a reminder to build can also be understood to promote the availability of building land in the municipality.

Numerous studies have shown that land policy interventions that restrict land supply from development lead to higher land costs (e.g., Pollakowski and Wachter, 1990; Barlow, 1993; Monk and Whitehead, 1996). Moreover, this can be argued to results in a lower supply of housing (Glaeser and Gyourko, 2003; Jackson, 2016). This finding is compatible with the basic idea of economic theory, which predicts that an imposition of a cost shifts the supply curve up by the amount of the cost, resulting in a lower quantity of new homes and a higher price paid by new homebuyers (e.g., Brueckner, 2009). Conversely, this implies that higher land supply results in higher housing supply (Grimes and Aitken, 2010; White and Allmendinger, 2003). This provides the understanding of the basic mechanism of how a land policy intervention in the form of a reminder to build is linked to the (in) responsiveness of housing supply and ultimately the quantity of housing supply (see also Fig. 2 in Section 2).

#### References

- Accetturo, A., Lamorgese, R., Mocetti, S., Pellegrino, D., 2021. Housing supply elasticity and growth: evidence from Italian cities. J. Econ. Geogr. 21 (3), 367–396.
- Albouy, D., Ehrlich, G., 2018. Housing productivity and the social cost of land-use restrictions. J. Urban Econ. 107, 101–120.
- Alexander, E.R., 1998. Planning and implementation: coordinative planning in practice. Int. Plan. Stud. 3, 303–320. https://doi.org/10.1080/13563479808721716.
- Alexander, F., 2008. Land Banking as Metropolitan Policy. Blueprint for American Prosperity. Available at SSRN: (https://ssrn.com/abstract=1955584).
- Alterman, R., 2012. Land-Use Regulations and Property Values: The < Windfalls Capture > Idea Revisited. Oxf. Handb. Urban Econ. Plan. 1–34.
- Anthony, J., 2003. The effects of Florida's growth management act on housing affordability. J. Am. Plann. Assoc. 69 (3), 282–295.
- Anthony, J., 2006. State growth management and housing prices. Soc. Sci. Q. 87 (1), 122–141.
- Anthony, J., 2017. Housing price effects of growth regulations: a concise taxonomy. Int. J. Hous. Policy 17 (4), 569–590 https://doi.org/10.1080/14616718.2016.1215963.
- Ball, M., 2003. Markets and the structure of the housebuilding industry: an international perspective. Urban Stud. 40, 897–916. https://doi.org/10.1080/ 0042098032000074236.
- Ball, M., 2011. Planning delay and the responsiveness of English housing supply. Urban Stud. 48, 349–362. https://doi.org/10.1177/0042098010363499.
- Barlow, J., 1993. Controlling the housing land market: some examples from Europe. Urban Stud. 30, 1129–1149. https://doi.org/10.1080/00420989320081091.
- Been, V., 2005. Impact fees and housing affordability. Cityscape 8, 139–185.Been, V., Ellen, I.G., O'Regan, K., 2019. Supply skepticism: housing supply and affordability. Hous. Policy Debate 29 (1), 25–40.
- Bento, A., Lowe, S., Knaap, G.J., Chakraborty, A., 2009. Housing market effects of inclusionary zoning. Cityscape 11, 7–26. https://doi.org/10.4324/9780203722671.
- Bovet, J., Reese, M., Köck, W., 2018. Taming expansive land use dynamics Sustainable land use regulation and urban sprawl in a comparative perspective. Land Use Policy 77. 837–845. https://doi.org/10.1016/i.landusepol.2017.03.024.
- Bramley, G., 1998. Measuring planning: Indicators of planning restraint and its impact on housing land supply. Environ. Plan. B Plan. Des. 25, 31–55. https://doi.org/ 10.1068/b250031.
- Bramley, G., Watkins, D., 2014. Measure twice, cut once'— revisiting the strength and impact of local planning regulation of housing development in England. Environ. Plan. B Plan. Des. 41, 863–884. https://doi.org/10.1068/b39131.
- Bratu, C., Harjunen, O., Saarimaa, T., 2023. JUE Insight: city-wide effects of new housing supply: evidence from moving chains. J. Urban Econ. 133, 103528 https://doi.org/ 10.1016/j.jue.2022.103528.
- Brueckner, J.K., 2000. Urban sprawl: diagnosis and remedies. Int. Reg. Sci. Rev. 23 (2), 160-171.
- Brueckner, J.K., 2009. Government land use interventions: an economic analysis. In: Lall, S.V., Freire, M., Yuen, B., Rajack, R., Helluin, J.-J. (Eds.), Urban Land Markets:

Improving Land Management for Successful Urbanization. Springer, New York, NY, pp. 3–23.

- Buitelaar, E., 2004. A transaction-cost analysis of the land development process. Urban Stud. 41 (13), 2539–2553.
- Buitelaar, E., 2010. Cracks in the myth: challenges to land policy in the Netherlands. Tijdschr. voor Econ. En. Soc. Geogr. 101, 349–356. https://doi.org/10.1111/j.1467-9663.2010.00604.x.
- Burge, G., Ihlanfeldt, K., 2006. Impact fees and single-family home construction. J. Urban Econ. 60 (2), 284–306.
- Caesar, C., 2016. Municipal land allocations: integrating planning and selection of developers while transferring public land for housing in Sweden. J. Hous. Built Environ. 31, 257–275. https://doi.org/10.1007/s10901-015-9457-2.
- Capozza, D., Li, Y., 1994. The intensity and timing of investment: the case of Land. Am. Econ. Rev. 84 (4), 889–904.
- Cheshire, P., Sheppard, S., 2002. The welfare economics of land use planning. J. Urban Econ. 52 (2), 242–269.
- Cheshire, P., Vermeulen, W., 2009. Land markets and their regulation: the welfare economics of planning. In: Geyer, H.S. (Ed.), International Handbook of Urban Policy, Vol. II: Issues in the Developed World. Edward Elgar, Cheltenham, pp. 151–193.
- Coiacetto, E., 2009. Industry structure in real estate development: Is city building competitive? Urban Policy Res 27, 117–135. https://doi.org/10.1080/ 08111140802499080.
- Cunningham, C., 2007. Growth Controls, Real Options, and Land Development. Rev. Econ. Stat. 89 (2), 343–358.
- Debrunner, G., Hartmann, T., 2020. Strategic use of land policy instruments for affordable housing - coping with social challenges under scarce land conditions in Swiss Cities. Land Use Policy 99, 104993. https://doi.org/10.1016/j. landusepol.2020.104993.
- van der Krabben, E., Jacobs, H.M., 2013. Public land development as a strategic tool for redevelopment: reflections on the Dutch experience. Land Use Policy 30, 774–783. https://doi.org/10.1016/j.landusepol.2012.06.002.
- Diamond, R., 2017. Housing supply elasticity and rent extraction by state and local governments. Am. Econ. J.: Econ. Policy 9 (1), 74–111.
- Evans, A.W., 2004. Economics and land use planning, Economics and Land Use Planning. Blackwell Publishing. https://doi.org/10.4324/9781315112022.
- Ganong, P., Shoag, D., 2017. Why has regional income convergence in the US declined? J. Urban Econ. 102, 76–90.
- Gerber, J.D., Hengstermann, A., Viallon, F., 2018b. Land Policy: how to deal with scarcity? In: Gerber, J.D., Hartmann, T., Hengstermann, A. (Eds.), Instruments of Land Policy - Dealing with Scarcity of Land. Routledge, pp. 8–26.
- Gerber, J.D., Hartmann, T., Hengstermann, A., 2018a. Planning with or against property rights. In: Gerber, J.D., Hartmann, T., Hengstermann, A. (Eds.), Instruments of Land Policy - Dealing with Scarcity of Land. Routledge, pp. 337–349.

Glaeser, E.L., Gyourko, J., 2003. The impact of building restrictions and housing affordability. Econ. Policy Rev. 21–39.

Glaeser, E.L., 2019. The macroeconomic implications of housing supply restrictions. Hot Prop. 99–108. https://doi.org/10.1007/978-3-030-11674-3\_8.

Glaeser, E.L., Ward, B.A., 2009. The causes and consequences of land use regulation: Evidence from Greater Boston. J. Urban Econ. 65, 265–278. https://doi.org/ 10.1016/j.jue.2008.06.003.

Glaeser, E.L., Gyourko, J., Saks, R.E., 2006. Urban growth and housing supply. J. Econ. Geogr. 6 (1), 71–89.

- Grossman, Z., Pincus, J., Shapiro, P., Yengin, D., 2019. Second-best mechanisms for land assembly and hold-out problems. J. Public Econ. 175, 1–16. https://doi.org/ 10.1016/j.jpubeco.2019.03.002.
- Gyourko, J., Molloy, R., 2015. Regulation and Housing supply. In: Duranton, G., Henderson, V., Strange, W. (Eds.), Handbook of Regional and Urban Economics, vol. 5. Elsevier, pp. 1289–1337.
- Gyourko, J., Saiz, A., 2004. Reinvestment in the housing stock: the role of construction costs and the supply side. J. Urban Econ. 55, 238–256. https://doi.org/10.1016/j. jue.2003.09.004.
- Gyourko, J., Saiz, A., Summers, A., 2008. A new measure of the local regulatory environment for housing markets: the Wharton residential land use regulatory index. Urban Stud. 45, 693–729. https://doi.org/10.1177/0042098007087341.
- Gyourko, J., Hartley, J.S., Krimmel, J., 2021. The local residential land use regulatory environment across U.S. housing markets: evidence from a new Wharton index. J. Urban Econ. 124, 103337 https://doi.org/10.1016/j.jue.2021.103337.
- Hannah, L., Kim, K.H., Mills, E.S., 1993. Land use controls and housing prices in Korea. Urban Stud. 30, 147–156. https://doi.org/10.1080/00420989320080091.
- Hartmann, T., Spit, T., 2015. Dilemmas of involvement in land management comparing an active (Dutch) and a passive (German) approach. Land Use Policy 42, 729–737. https://doi.org/10.1016/j.landusepol.2014.10.004.
- Henderson, V., 2009. The effect of residential land market regulations on urban welfare. In: Lall, S.V., Freire, M., Yuen, B., Rajack, R., Helluin, J.-J. (Eds.), Urban Land Markets: Improving Land Management for Successful Urbanization. Springer, New York, NY, pp. 25–49.
- Hengstermann, A., 2018. Building obligations in Switzerland: Overcoming the passivity of plan implementation. In: Gerber, J.D., Hartmann, T., Hengstermann, A. (Eds.), Instruments of Land Policy - Dealing with Scarcity of Land. Routledge, pp. 175–188.

Hilber, C.A.L., Vermeulen, W., 2016. The impact of supply constraints on house prices in England. Econ. J. 126, 358–405. https://doi.org/10.1111/ecoj.12213.

- Hsieh, C.-T., Moretti, E., 2019. Housing constraints and spatial misallocation. Am. Econ. J. 11 (2), 1–39.
- Huffman, F.E., Nelson, A.C., Smith, M.T., Stegman, M.A., 1988. Who bears the burden of development impact fees? J. Am. Plann. Assoc. 54, 49–55.
- Ihlanfeldt, K.R., 2004. Exclusionary Land-use Regulations within Suburban
- Communities: A Review of the Evidence and Policy Prescriptions. Urban Stud. 41, 261–283.
- Ihlanfeldt, K.R., 2007. The effect of land use regulation on housing and land prices. J. Urban Econ. 61, 420–435. https://doi.org/10.1016/j.jue.2006.09.003.
- Ihlanfeldt, K.R., Shaughnessy, T.M., 2004. An empirical investigation of the effects of impact fees on housing and land markets. Reg. Sci. Urban Econ. 34, 639–661. https://doi.org/10.1016/j.regsciurbeco.2003.11.002.

Irwin, E., Bockstael, N., 2004. Land use externalities, open space preservation, and urban sprawl. Reg. Sci. Urban Econ. 34 (6), 705–725.

- Jackson, K., 2016. Do land use regulations stifle residential development? Evidence from California cities. J. Urban Econ. 91, 45–56. https://doi.org/10.1016/j. inc.2015.11.004.
- Jackson, K., 2018. Regulation, land constraints, and California's boom and bust. Reg. Sci. Irban Econ. 68, 130–147. https://doi.org/10.1016/j.regsciurbeco.2017.10.005
- Urban Econ. 68, 130–147. https://doi.org/10.1016/j.regsciurbeco.2017.10.005. Kok, N., Monkkonen, P., Quigley, J., 2014. Land use regulations and the value of land and housing: An intra-metropolitan analysis. J. Urban Econ. 81, 136–148. https:// doi.org/10.1016/j.jue.2014.03.004.
- Krigsholm, P., Puustinen, T., Falkenbach, H., 2022. Understanding variation in municipal land policy strategies: An empirical typology. Cities 126, 103710. https://doi.org/ 10.1016/j.cities.2022.103710.
- Land Use and Building Act, 132 Parliament of Finland (1999, amendments up to 1211/2020).

Levine, N., 1999. Eff. Local Growth Controls Reg. Hous. Prod. Popul. Redistrib. Calif. 36, 2047–2068.

- Lewis, P.G., Marantz, N.J., 2019. J. Am. Plan. Assoc. What Plan. Know Using Surv. Local Land Use Regul. Under Hous. Dev. View Suppl. Mater. https://doi.org/10.1080/ 01944363.2019.1643253.
- Listokin, D., Hattis, D.B., 2005. Building Codes and Housing. City A J. Policy Dev. Res. 8, 21–67.
- Lyytikäinen, T., 2009. Three-rate property taxation and housing construction. J. Urban Econ. 65 (3), 305–313.
- Maestas, C., Buttice, M., Stone, W., 2014. Extracting wisdom from experts and small crowds: Strategies for improving informant-based measures of political concepts. Political Anal. 22 (3), 354–373.
- Mast, E., 2021. JUE insight: the effect of new market-rate housing construction on the low- income housing market. J. Urban Econ. 133, 103383 https://doi.org/10.1016/ j.jue.2021.103383.

May, P., 2003. Performance-based regulation and regulatory regimes: The saga of leaky buildings. Law Policy 25 (4), 381–401.

- Mayer, C.J., Somerville, C.T., 2000. Land use regulation and new construction. Reg. Sci. Urban Econ. 30, 639–662. https://doi.org/10.1016/S0166-0462(00)00055-7.
- Mayo, S., Sheppard, S., 1996. Housing supply under rapid economic growth and varying regulatory stringency: an international comparison. J. Hous. Econ. 5 (3), 274–289.

McFarlane, A., 1999. Taxes, fees, and urban development. J. Urban Econ. 46, 416–436. https://doi.org/10.1006/juec.1999.2130.

McFarlane, A., Li, J., Hollar, M., 2021. Building codes: what are they good for? Cityscape 23 (1), 101–132. Metcalf, G., 2018. Sand castles before the tide? affordable housing in expensive cities. J. Econ. Perspect. 32 (1), 59–80. https://doi.org/10.1257/jep.32.1.59.

Mettepenningen, E., Beckmann, E., Eggers, J., 2011. Public transaction costs of agrienvironmental schemes and their determinants - analysing stakeholders' involvement and perceptions. Ecol. Econ. 70 (4), 641–650. https://doi.org/10.1016/ j.ecolecon.2010.10.007.

Miceli, T., Sirmans, C.F., 2007. The holdout problem, urban sprawl, and eminent domain. J. Hous. Econ. 16 (3-4), 309–319. https://doi.org/10.1016/j. ihe.2007.06.004.

Miles, M.B., Huberman, A.M., 1994. Qualitative data analysis: An Expanded Sourcebook. Sage, Thousand Oaks, CA.

Monk, S., Whitehead, C.M.E., 1996. Land supply and housing: a case-study. Hous. Stud. 11, 407–423. https://doi.org/10.1080/02673039608720865.

Muñoz Gielen, D., Maguregui Salas, I., Burón Cuadrado, J., 2017. International comparison of the changing dynamics of governance approaches to land development and their results for public value capture. Cities 71, 123–134. https:// doi.org/10.1016/j.cities.2017.05.012.

Murray, C.K., 2018. Developers pay developer charges. Cities 74, 1–6. https://doi.org/ 10.1016/j.cities.2017.10.019.

Murray, C.K., 2020. Time is money: How landbanking constrains housing supply. J. Hous. Econ. 49, 101708 https://doi.org/10.1016/j.jhe.2020.101708.

- Murray, C.K., 2022. A housing supply absorption rate equation. J. Real. Estate Fin. Econ. 64 (2), 228–246. https://doi.org/10.1007/s11146-020-09815-z.
- Nahrath, S., 2018. A Swiss perspective on pre-emption rights: impact without application. In: Gerber, J.D., Hartmann, T., Hengstermann, A. (Eds.), Instruments of Land Policy - Dealing with Scarcity of Land. Routledge, pp. 213–217.
- Needham, B., 1997. Land policy in the Netherlands. Tijdschr. voor Econ. En. Soc. Geogr. 88, 291–296.
- Needham, B., 2000. Land taxation, development charges, and the effects of land-use. J. Prop. Res. 17 (3), 241-257.
- Nelson, A., 1994. Development impact fees: the next generation. Urban Lawyer 26 (3), 541–562.
- Nelson, A., Dawkins, C., Sanchez, T., 2004. Urban containment and residential segregation: a preliminary investigation. Urban Stud. 41 (2), 423–439.

NLS, 2020. Statistical information on real estate transactions [e-publication]. [Cited 10.3.2023]. Available: <a href="https://khr.maanmittauslaitos">https://khr.maanmittauslaitos</a>.

fi/tilastopalvelu/rest/API/kiinteistokauppojen-tilastopalvelu.html?lang=en#).
OECD, 2008. Handbook on Constructing Composite Indicators: Methodology and User Guide. OECD Publishing, Paris. https://doi.org/10.1787/9789264043466-en.

Oikarinen, E., Bourassa, S., Hoesli, M., Engblom, J., 2023. Revisiting metropolitan house price-income relationships. J. Hous. Econ. 61, 101946 https://doi.org/10.1016/j. jhe.2023.101946.

Paciorek, A., 2013. Supply constraints and housing market dynamics. J. Urban Econ. 77, 11–26. https://doi.org/10.1016/j.jue.2013.04.001.

Pendall, R., 2020. Knowing what land use regulations localities have "on the books" can reveal regulatory stringency—and much more. J. Am. Plan. Assoc. 86 (2), 264–265.

- Pollakowski, H.O., Wachter, S.M., 1990. The effects of land-use constraints on housing prices. Land Econ. 66 (3), 315–324. https://doi.org/10.2307/3146732.
- Powell, B., Stringham, E., 2004. Housing supply and affordability: Do affordable housing mandates work? Los Angeles, CA.

Quigley, J., Rosenthal, L.A., 2005. The effects of land-use regulation on the price of housing: what do we know? what can we learn? Cityscape 8, 69–137.

Rinkinen, K., Kinnunen, J., 2017. Asemakaavoituksen muutokset - Kaavojen määrän, keston ja kaavoilla tuotetun kerrosalan vertailu. Ympäristöministeriön raportteja 20/2017. Available: (http://julkaisut.valtioneuvosto.fi/handle/10024/80531).

Robson, C., 2002. Real World Research, 2nd ed. Blackwell Publishers Inc, Malden, USA.

Saks, R.E., 2008. Job creation and housing construction: constraints on metropolitan area employment growth. J. Urban Econ. 64 (1), 178–195.

- Schill, M.H., 2005. Regulations and housing development: what we know. Cityscape 8, 5–19. https://doi.org/10.4324/9780203722671.
- Schuetz, J., Meltzer, R., Been, V., 2011. Silver Bullet or Trojan Horse? the effects of inclusionary zoning on local housing markets in the United States. Urban Stud. 48 (2), 297–329. https://doi.org/10.1177/0042098009360683.

Shahab, S., Clinch, J., O'Neill, E., 2019. Impact-based planning evaluation: advancing normative criteria for policy analysis. Environ. Plan. B: Urban Anal. City Sci. 46 (3), 534–550.

- Shahab, S., Hartmann, T., Jonkman, A., 2021. Strategies of municipal land policies: Housing development in Germany, Belgium, and Netherlands. Eur. Plan. Stud. 29 (6), 1132–1150. https://doi.org/10.1080/09654313.2020.1817867.
- Skidmore, M., Peddle, M., 1998. Do development impact fees reduce the rate of residential development? Growth Change 29, 383–400. https://doi.org/10.1111/ j.1468-2257.1998.tb00026.x.
- Staley, S.R., 2001. Ballot-box zoning, transaction costs, and urban growth. J. Am. Plan. Assoc. 67, 25–37. https://doi.org/10.1080/01944360108976353.

Tappendorf, J., Denzin, B., 2011. Turning vacant properties into community assets through land banking. Urban Lawyer 43 (3), 801–812.

- Titman, S., 1985. Urban land prices under uncertainty. Am. Econ. Rev. 75 (3), 505-514.
- Triantafyllopoulos, N., 2017. On the origins of tourist urbanisation in Greece: Land speculation and property market (in)efficiency. Land Use Policy 68, 15–27. https:// doi.org/10.1016/j.landusepol.2017.06.007.
- Tse, R.Y.C., 1998. Housing price, land supply and revenue from land sales. Urban Stud. 35, 1377–1392. https://doi.org/10.1080/0042098984411.
- Turnbull, G.K., 1988. Property taxes and the transition of land to urban use. J. Real. Estate Financ. Econ. 1, 393–403. https://doi.org/10.1007/BF00187074.
- Turner, M., Haughwout, A., Van Der Klaauw, W., 2014. Land Use Regulation and Welfare. Econometrica 82, 1341–1403. https://doi.org/10.3982/ecta9823.

- Valtonen, E., Falkenbach, H., Viitanen, K., 2017b. Development-led planning practices in a plan-led planning system: empirical evidence from Finland. Eur. Plan. Stud. 25, 1053–1075. https://doi.org/10.1080/09654313.2017.1301885.
- Valtonen, E., Falkenbach, H., van der Krabben, E., 2017a. Risk management in public land development projects: comparative case study in Finland, and the Netherlands. Land Use Policy 62, 246–257. https://doi.org/10.1016/j.landusepol.2016.12.016.
- Valtonen, E., Falkenbach, H., Vittanen, K., 2018. Securing public objectives in largescale urban development: comparison of public and private land development. Land Use Policy 78, 481–492. https://doi.org/10.1016/j.landusepol.2018.07.023.
- Vejchodská, E., Shahab, S., Hartmann, T., 2022. Revisiting the Purpose of Land Policy: Efficiency and Equity. J. Plan. Lit. 37 (4), 575–588.
- Viitanen, K., Palmu, J., Kasso, M., Hakkarainen, E., Falkenbach, H., 2003. Real estate in Finland. Helsinki University of Technology, Institute of Real Estate Studies, Espoo, p. 68. ISBN 951-22-6875-2.
- Wang, C., 2014. Regulating land development in a natural disaster-prone area: the roles of building codes. Resour. Energy Econ. 36 (1), 209–228. https://doi.org/10.1016/j. reseneeco.2012.10.002.

- Wang, H., Wu, X., Wu, D., Nie, X., 2019. Will land development time restriction reduce land price? the perspective of American call options. Land Use Policy 83, 75–83. https://doi.org/10.1016/j.landusepol.2019.01.028.
- White, M., Allmendinger, P., 2003. Land-use planning and the housing market the UK and the USA. Urban Stud. 40 (5-6), 953–972.
- Wong, F., Chan, E., Yu, A., 2011. Property developers' major cost concerns arising from planning regulations under a high land-price policy. J. Urban Plan. Dev. 137 (2), 112–120.
- Wrenn, D., Irwin, E., 2015. Time is money: an empirical examination of the effects of regulatory delay on residential subdivision development. Reg. Sci. Urban Econ. 51, 25–36. https://doi.org/10.1016/j.regsciurbeco.2014.12.004.
- Zabel, J.E., 2012. Migration, housing market, and labor market responses to employment shocks. J. Urban Econ. 72 (2–3), 267–284.
- Zabel, J.E., Paterson, R.W., 2006. The effects of critical habitat designation on housing supply: an analysis of California housing construction activity. J. Reg. Sci. 46, 67–95. https://doi.org/10.1111/j.0022-4146.2006.00433.x.